# ON THE EVOLUTION OF MARKET INSTITUTIONS: THE PLATFORM DESIGN PARADOX

# CARLOS ALÓS-FERRER GEORG KIRCHSTEIGER MARKUS WALZL

# CESIFO WORKING PAPER NO. 2012 CATEGORY 9: INDUSTRIAL ORGANISATION JUNE 2007

Presented at CESifo Area Conference on Applied Microeconomics, March 2007

An electronic version of the paper may be downloaded• from the SSRN website:http://www.SSRN.com/abstract=993675• from the RePEc website:www.RePEc.org• from the CESifo website:www.CESifo-group.de

# ON THE EVOLUTION OF MARKET INSTITUTIONS: THE PLATFORM DESIGN PARADOX

# Abstract

We study competition among market designers who create new trading platforms, when boundedly rational traders learn to select among them. We ask whether efficient platforms, leading to market - clearing trading outcomes, will dominate the market in the long run. If several market designers are competing, we find that traders learn to select non-market clearing platforms with prices systematically above the market-clearing level, provided at least one such platform is introduced by a market designer. This in turn leads market designers to introduce non-market clearing platforms. Hence platform competition induces non-competitive market outcomes.

JEL Code: C72, D4, D83, L1.

Keywords: market institutions, evolution of trading platforms, learning, asymmetric rationality.

Carlos Alós-Ferrer Department of Economics University of Konstanz BoxD-150 78457 Konstanz Germany Carlos.Alos-Ferrer@uni-konstanz.de Georg Kirchsteiger ECARES Free University of Brussels Avenue F D Roosevelt 50, CP114 1050 Brussels Belgium gkirchst@ulb.ac.be

Markus Walzl Department of Economics Maastricht University P.O. Box 616 6200 MD Maastricht The Netherlands m.walzl@algec.unimaas.nl

We are grateful to Dan Friedman and seminar participants in Berlin, UL Brussels, University College London, Louvain-la-Neuve and Maastricht for helpful comments. The first author acknowledges funding by the Austrian Science Fund (FWF) under Project P18141-G09, the second author by the Banque Nationale de Belgique, and the third author by the Dutch Science Foundation (NWO).

#### 1. INTRODUCTION

Markets are not only characterized by demand and supply, but also by the rules that govern the trading process. The "institutional" framework determines the set of market participants, their available options, and the matching and information structure of the market. In reality we observe a huge variety of different market frameworks even for trading the very same good. Real estate, for example, is traded at auctions as well as by personal bargaining. There is also a large amount of evidence (mainly experimental) that these characteristics are crucial for the resulting trading outcome and for the realized prices (for an overview see e.g. Plott [33] or Holt [23]; in the context of financial markets see also Friedman [18]). While double auctions typically tend to generate market clearing prices and quantities, posted offer markets establish prices that tend to be above the market clearing level, whereas the prices on posted bid markets seem to be below the Walrasian level (see e.g. Plott and Smith[34]). As a consequence, some gains of trade are not realized on these trading platforms, and inefficiencies occur due to the design of the trading platform. In a similar way, Dutch or first-price auctions are notorious for inducing overbidding and creating inefficient allocations compared to second-price formats (see e.g. Kagel [25]). In a field study, Roth and Ockenfels [37] show that fixed ending-rules ("hard-close") in online auctions lead to late bidding ("sniping"); see also [32]. A laboratory experiment by Ariely, Ockenfels, and Roth [3] confirmed this finding, and also showed that fixed ending-rules lead to lower revenues for the seller (and less efficient allocations) than automatic extensions of the auction ("soft ending"). All these studies suggest that socially desirable features of market outcomes such as unbiased (market-clearing) prices and efficient allocations are rather sensitive to details of the respective market institution. Moreover, there seems to be a trade-off between efficiency and a price-bias in favor of one of the market-sides.

A remarkable example for the coexistence of a variety of trade-institutions is provided by Business to Business (B2B) trading platforms (for a comprehensive economic analysis see e.g. Lucking-Reiley and Spulber [30]). The last decades have seen a proliferation of B2B platforms, and despite the burst of the internet bubble there were more than 1000 B2B marketplaces active in Europe in 2003 (see European Commission [17]). While most of the public and scientific attention is devoted to e-marketplaces targeting consumers (like e-bay or Yahoo), about 95% of the ecommerce is actually B2B (see United Nations [42]). In 2004 B2B had an estimated volume of \$1 trillion (see The Economist [11]). Contrary to Business to Consumers or Consumer to Consumer platforms, large quantities of relatively standardized products are traded at B2B exchanges. On these platforms agents seem to act either as buyers or as sellers, but not as both (see European Commission [17]). B2B ecommerce is basically organized in three different ways. The predominant modus in the early days of e-commerce were platforms opened by buyers or sellers (or respective umbrella organizations). An example is MetalSite, a platform organized by steel producers that suspended operations in 2001. Currently, B2B e-commerce is typically organized either as e-procurement<sup>1</sup> where sellers use standardized software and exchange opportunities offered by platforms such as Ariba or CommerceOne to design and allocate procurement contracts. Or trade is organized via institutions operated by a third party (this holds e.g. for CheMatch - a trade platform for chemical products - or for a large part of the product portfolio offered at EnronOnline a multi–commodity exchange run by Enron until 2002). Of all firms active on B2B platforms, about one third operates on such platforms run by third parties (see European Commission [17]). Both e-procurement software and market designs of third parties show a variety of institutional arrangements. EnronOnline, for instance, was organized as a posted offer market while competing platforms such as AltraEnergy (or on the chemical sector CheMatch) are exchange platforms that work like double auctions. The software solutions offered by Ariba and CommerceOne include various institutional arrangements such as Dutch auctions or proxy–bidding (with hard and soft ending rules). Hence, platform designs that show different efficiency properties in the laboratory co-exist in B2B e-commerce.

Given the variety of different market institutions, and the variety of their efficiency properties, one wonders which type(s) of trading institutions will be observed in the long run. In particular, our paper investigates whether institutions promoting efficient, market clearing outcomes will dominate less efficient trading platforms in the long run. To answer this question, we have to investigate the evolution of market institutions. It is useful to distinguish between two aspects of this evolution, namely the selection between existing institutions by the traders, and the emergence of new institutions. New market institutions can either be introduced on purpose by a market designer, or be the (unintended) by-product of the actions of the traders. In what follows we will focus on market platforms introduced on purpose.<sup>2</sup> If a trading platform is introduced by a (profit maximizing) market designer who demands user fees, the design of a new platform by the designer and the selection among existing ones by the traders are closely interlinked. The market designer will try to introduce a new platform with characteristics that attract many traders. This attractiveness in turn determines the long run survival of the platform. In this paper we analyze this interplay between the creation of new and the selection among existing trading platforms, and we investigate the characteristics of the resulting platforms with respect to their ability to achieve market clearing outcomes.

Trading platforms are created by profit maximizing and risk-neutral market designers. The designers compete with each other through platform designs. Each designer chooses a trading fee that he demands from the traders for the use of his platform. To capture the trade-off between efficiency and a price-bias for one market side that has been observed in laboratory studies (see above), we allow that the designers can choose a platform design with a systematic price-bias, above or below

<sup>&</sup>lt;sup>1</sup>For a recent discussion of the adoption of e-procurement in B2B and an overview of market designs see e.g. Davila, Gupta, and Palmer [10]

<sup>&</sup>lt;sup>2</sup>For an analysis of markets as a by-product of traders' actions, see Kirchsteiger et al.[28].

the market clearing price. Hence, by the trading fee each designer decides upon his share of the surplus created through trade at his platform. But he can also favor one type of trader with the introduction of a price bias. Any bias reduces the surplus generated at the platform (and thereby ceteris paribus the revenue for the designer), but may also make it more attractive for the favored type of trader, which in turn may enhance the platform's survival probability.

To analyze this trade-off, we model competition between two market designers and compare the results of this setting with the benchmark case of a monopolistic market designer. After the platforms have been designed, each trader decides on which platform to be active (for the monopolistic case, there is of course no real choice traders just trade at the only existing platform). The role of the trader (buyer or seller) is exogenously given. Sellers are assumed to be firms with a constant returns to scale production technology.<sup>3</sup> Buyers are characterized by their demand functions, and might be either consumers or other firms. For given platform characteristics, the selection by traders gives rise to a coordination game. If each trader opts for a particular platform no trader has an incentive to deviate from this platform - independently of the design alternatives offered by the competing platform. However, traders might learn to coordinate on a particular platform. Following the game-theoretic learning literature (see Young [43], Kandori et al. [26], or Ellison [13]), we use a Markovian model to analyze the platform choice of the traders. We assume that the traders' behavior depends on the market outcomes generated by the different platforms and thereby on the characteristics of all feasible platforms. We are interested in the long term properties of this learning process, i.e. in its limit invariant distribution. This limit invariant distribution in turn determines the payoffs of the market designers. Hence, we establish a link between designer-revenues and the characteristics of all feasible platforms.

For the case of competing platforms we find that - in the long run - traders will always coordinate on a platform with prices above the market clearing level, provided that such a platform has been introduced by at least one designer. This forces both designers to introduce platforms that are not market clearing, but that have a price bias in favor of the sellers. On the other hand we find that a monopolistic designer will always introduce a market clearing platform. Therefore competition at the designers' level turns out to be detrimental for a competitive outcome at the traders' level. We regard this result as paradoxical.

The present paper is related to three strands of the literature. First, since we investigate the role of trading platforms with exogenously given buyers and sellers, our paper is to some extent related to the two-sided markets literature (see Rochet and Tirole [36] for an overview). This literature is based on the assumption of network-externalities. It analyzes the impact of these externalities and of platform competition on the structure of the fees demanded by the market designers (see e.g. Armstrong

 $<sup>^{3}\</sup>mathrm{In}$  Appendix B we investigate the robustness of our results with respect to decreasing returns to scale.

[4], Belleflamme and Toulemonde [5], Caillaud and Julien [9], or Rochet and Tirole [35]). In contrast, we want to investigate whether traders learn to coordinate on market clearing trading platforms, if such platforms are feasible. Therefore we explicitly model the learning behavior of the traders, whereas the two-sided market literature assumes rational traders. Further, we ask whether platform competition induces market designers to establish platforms with characteristics that achieve market clearing outcomes. Consequently, we abstract from any network externalities that are not internalized by the price at which trade takes place. In our model trading fees demanded by the market designers are neutral insofar as the market outcome is only influenced by the total fee imposed on both market sides, but not on the distribution of the fees on the two market sides.<sup>4</sup>

Second, our paper is also related to the literature on competition between exogenously given trading institutions. Ellison and Fudenberg [15] and Ellison et al. [16] analyze under which circumstances different market institutions can coexist in equilibrium. Due to their different research questions these papers do not allow for institutions with systematic price biases. Kugler et al. [29] and Neeman and Vulkan [31] investigate the case of centralized versus decentralized trading institutions. All of these papers they rely on the assumption of rational traders, and do not allow for learning. In terms of traders' behavior, the learning model of Gerber and Bettzüge [22] is relatively close to our paper. But since they focus in the possibility of multiplicity of active trading platforms, they consider neither non-market-clearing platforms nor market designers. Closest related to the paper at hand is the work by Alós-Ferrer and Kirchsteiger [2], which also analyzes the learning behavior of traders who face the choice between different, not necessarily market clearing platforms. That paper, however, deals only with the selection among different, exogenously given institutions and does not consider competition between market designers.

In our model, rational market designers are confronted with boundedly rational, learning traders.<sup>5</sup> Hence, our paper belongs to a small but growing literature that we would like to call "asymmetric rationality," where fully rational firms or otherwise sophisticated agents are confronted with a population of boundedly rational ones. The basic motivation is that consumers and small traders do not have the resources to obtain all the relevant information and fully optimize their behavior, often relying on behavioral rules of thumb instead. However, large firms, market designers, etc. can be taken as comparatively sophisticated. Schlag [40], Gabaix and Laibson [21], Hopkins [24], and Spiegler [41] apply this approach to the analysis of industries facing boundedly rational consumers. See Ellison [14] for an overview of this literature.

The paper is organized as follows. Section 2 presents the basic model. Section 3 discusses the traders' platform choice of the traders. Section 4 analyzes the design of the platform. Section 5 concludes. All proofs not found in the main text are

 $<sup>^{4}</sup>$ Rochet and Tirole [36] define two-sided markets by the non-neutrality of the fees. In their terminology we model a one-sided market.

<sup>&</sup>lt;sup>5</sup>In Appendix B we show the robustness of our results with respect to learning designers.

in Appendix A. In Appendix B we analyze the robustness of our results with respect to boundedly rational designers and with respect to decreasing returns to scale production. Appendix C provides a short reference for Markov processes.

## 2. The model

**2.1.** Market platforms' design. A homogenous good is traded at alternative market platforms, which are set up by market designers. For simplicity, we restrict our attention to two competing market designers (referred to as competitive market design). As a benchmark, we will also analyze the case where only one market designer can set up a trading platform (referred to as monopolistic market design).

Before any trade takes place, the market designers decide upon the set of trading rules under which their respective platforms operate, and the trading fees they demand from the traders. We do not aim at a complete description of the different sets of rules the designers can introduce. Rather, we characterize them by their ability to establish market clearing. Market designers may choose to design platforms such that market clearing is guaranteed, or they may pick platforms where the price is systematically biased above or below the market clearing price. Denote by  $p_i^*$  the market clearing price if at least one seller and at least one buyer choose this platform and by  $\beta_i > 0$  the bias of platform i = 1, 2. The actual price at which trade takes place at platform i is then given by  $p_i = \beta_i p_i^*$ . If the actual price is not market clearing (i.e.  $\beta_i \neq 1$ ), the quantity traded is determined by the short market side, and traders on the long market side are rationed. Sellers are rationed equally if  $\beta_i > 1$ . We do not specify any rationing rule for the buyers.

The common set of feasible biases is assumed to be a finite, regular grid  $B = \{\beta_{\min}, \beta_{\min} + \delta, ..., 1, ... \beta_{\max} - \delta, \beta_{\max}\}$ , where  $0 < \beta_{\min} < 1 < \beta_{\max}$  and  $\delta$  is the step of the grid. To understand why institutions with different price biases are feasible for the designers, recall the experimental and empirical results mentioned in the introduction. In our framework posted offer markets or first price auctions are characterized by  $\beta > 1$ , posted bid markets or proxy-auctions with "hard–close" by a  $\beta < 1$ , while double auctions can be represented by  $\beta = 1$ . We refer to the platform with  $\beta = 1$  as the market-clearing platform, and we assume that such a platform is always feasible. |B| denotes the number of feasible biases.

After the platforms are set up, traders will use their experience and observations to eventually learn which platform to use. Formally, we will analyze a learning process with an infinite number of trading rounds. The designers' long-run payoffs are the expected per round charges. Furthermore, we assume that the charges of designer iare a fixed share of the revenue generated by trade on i's platform.<sup>6</sup> Denote by  $f_i$ the trading fee demanded by designer i, and by  $ER_i$  the expected per round revenue generated on platform i. Then market designer i's profits are given by  $\pi_{D,i} = f_i ER_i$ . The set of feasible fees is the same for both designers. For simplicity we assume

 $<sup>^{6}\</sup>mathrm{Our}$  results would not change if we assume quantity-dependent charges instead of revenue-dependent charges.

that it is given by a finite, regular grid  $F = \{f_{\min}, f_{\min} + \gamma, \dots f_{\max} - \gamma, f_{\max}\}$ , where  $0 < f_{\min} < f_{\max} < 1.^7 |F|$  denotes the number of feasible fees.

The trading fee can be imposed on the sellers' side, on the buyers' side, or divided between both sides. However, the market clearing price, the realized price at which trade is conducted, and the traded quantities depend only on the total fee, and not on the distribution of the fee over the two market sides. Buyers at platform i pay  $p_i$ for each unit, market designers receive  $f_i p_i$ , and sellers ultimately receive  $(1 - f_i) p_i$ . Hence, we do not need to specify on which market side the fee is imposed.

The characteristics of a platform i are denoted by  $s_i = (\beta_i, f_i)$ , and the set of feasible characteristics by  $S = B \times F$ .

**2.2.** Traders. The good is supplied by a finite set M of profit maximizing firms that use a constant returns to scale technology. We assume that there are at least two sellers at the market, i.e. |M| > 1. Each seller uses the same constant returns to scale technology with marginal costs of  $c > 0.^8$  When deciding about the supply, a seller takes into account the trading fee of the platform at which he is operating. Hence, sellers will supply a strictly positive but finite quantity if and only if the price net of trading fee is equal to c.

As we will see, the assumption of a constant returns to scale technology allows us to derive results for a very general class of learning models. That is, by focusing on the constant returns to scale case, we will obtain results that are robust to the details of the learning process. In the appendix we illustrate that for strictly decreasing returns to scale the results depend on the details of the learning model. In particular, the results of the constant returns to scale case can be replicated also for strictly decreasing returns to scale, but not for the whole class of learning models we analyze here.

The good is demanded by a finite set N of buyers with |N| > 1. Each buyer  $n \in N$ is endowed with a demand function  $d_n(p)$  which might be different for different buyers. All the demand functions are assumed to be strictly decreasing in p. Furthermore,  $0 < d_n(p) < \infty$  for all p, n. To avoid discontinuities in the designers' profit functions we also assume that  $\lim_{p\to\infty} pd_n(p) = 0$  for all  $n \in N$ .<sup>9</sup>

We call a platform *active* if both sellers and buyers are present and positive quantities are traded, and *inactive* if not. The presence of both types of traders does not ensure that the platform is active. Due to the assumption of a constant returns to scale technology the market clearing price of a platform *i* where both sellers and buyers are present (and, in particular, of an active platform) is given by  $p_i^*(s_i) = \frac{c}{1-t_i}$ .

<sup>&</sup>lt;sup>7</sup>The assumption that the fees are strictly positive can be justified by (unmodelled) setup costs for the market designers.

<sup>&</sup>lt;sup>8</sup>The assumption of identical sellers might seem restrictive at the first sight. Within our framework firms without access to the lowest cost technology cannot sell anything on the market, anyhow. Hence, our assumptions only rule out the case where exactly one firm has access to the production technology with the lowest costs.

<sup>&</sup>lt;sup>9</sup>Our results do not depend on the assumption that the value of demand goes to zero when the price approaches infinity. However, the presentation is simplified by this assumption.

The realized price at which trade is conducted on platform i is then

$$p_i(s_i) = \beta_i \frac{c}{1 - f_i}.$$

Note that if  $\beta_i < 1$ , the net price received by the sellers is below the marginal costs. Hence, supply is zero, and platform *i* is inactive.

Denote by  $N_i$  the set of buyers who choose platform i, and by  $M_i$  the set of sellers who choose platform i. Then, platform i is active if and only if  $|N_i| > 0$ ,  $|M_i| > 0$ , and  $\beta_i \ge 1$ . Let

$$D_{N_i}(p) = \sum_{n \in N_i} d_n(p)$$

denote the total demand at platform *i*. The quantities traded by a buyer  $n \in N_i$ ,  $q_{n,i}(N_i, M_i, s_i)$ , and by a seller  $m \in M_i$ ,  $q_{m,i}(N_i, M_i, s_i)$ , are given by

$$q_{n,i}(N_i, M_i, s_i) = \begin{cases} d_n(\beta_i \frac{c}{1-f_i}) & \text{if } i \text{ is active,} \\ 0 & \text{otherwise,} \end{cases}$$
$$q_{m,i}(N_i, M_i, s_i) = \begin{cases} \frac{1}{|M_i|} D_{N_i} \left(\beta_i \frac{c}{1-f_i}\right) & \text{if } i \text{ is active,} \\ 0 & \text{otherwise.} \end{cases}$$

In the single-designer case, traders cannot choose between different platforms, but have to use platform *i*. Hence,  $N_i = N$ ,  $M_i = M$ , and the market outcome is only determined by the platform characteristics  $s_i$ .

If there is competition between market designers, trade can take place at different platforms, and the outcome depends also on the way traders learn which platform to use. This learning process is driven by the market outcomes of both platforms (see above), and by the individual evaluations of these outcomes. For the latter part note that if *buyers* trade strictly positive amounts, they are strictly better off than without trade. Hence, inactive platforms are worse for buyers than active ones. Furthermore, whenever a buyer trades a strictly positive quantity, he is not rationed at all. Hence, it is natural to assume that buyers' evaluation of active platforms is monotonically decreasing in the price. Therefore, buyers' evaluation of platform i could be represented e.g. by<sup>10</sup>

$$\pi_{n,i}(s_i) = \begin{cases} \frac{1}{p_i} = \frac{1-f_i}{\beta_i c} & \text{if } i \text{ is active,} \\ 0 & \text{otherwise} \end{cases}$$

If both platforms i and j are active (i.e. positive amounts are traded),

$$p(s_i) < p(s_j) \Longleftrightarrow \pi_{n,i}(s_i) > \pi_{n,j}(s_j).$$

 $<sup>^{10}</sup>$ We do not use this particular representation. If demand is derived from utility maximization, though, the realized (indirect) utility must be (a strictly monotone transformation of) this payoff function.

Clearly, platform *i* is active if and only if both types of traders are present and  $\beta_i \geq 1$ . Both platforms are active whenever  $N_i, N_j \neq N, M_i, M_j \neq M$  and  $\beta_i, \beta_j \geq 1$ . In particular, if  $\beta_i = \beta_j = 1$  and  $f_i < f_j$ , then  $\pi_{n,i}(s_i) > \pi_{n,j}(s_j)$ .

The sellers' evaluation of the platforms are determined by the respective profits. An inactive platform gives of course zero profits. Furthermore, whenever  $\beta_i > 1$ , sellers trading on platform *i* are on the long market side, and hence equally rationed. Hence, the sellers' evaluation of platform *i* is given by

$$\pi_{m,i}(N_i, M_i, s_i) = \begin{cases} \left[\frac{1}{|M_i|} D_{N_i}\left(\beta_i \frac{c}{1-f_i}\right)\right] (\beta_i - 1)c & \text{if } i \text{ is active,} \\ 0 & \text{otherwise.} \end{cases}$$

Note that as long as both platforms are active, the outcome of a non-market clearing platform is always better for the sellers than the outcome of a market clearing platform. That is, for all  $f_i, f_j$ ,

$$N_i, N_j \neq N, M_i, M_j \neq M \text{ and } \beta_i > \beta_j = 1 \Rightarrow \pi_{m,i}(N_i, M_i, s_i) > \pi_{m,j}(N_i, M_i, s_i)$$
.

#### 3. The Traders' Platform Choice

First, market designers choose their platforms' characteristics. If there is only one market designer, traders' choice is trivial - they simply opt for the existing platform. With more than one market designer, traders have to choose between the two platforms. For any given  $s_i, s_j$  the choice of platform constitutes a coordination game. If all traders choose platform i, no trader has an incentive to deviate to the other platform j. Furthermore, if  $\beta_i$  and  $\beta_j$  are strictly larger than 1, full coordination on any platform is even a strict Nash equilibrium. Hence, nothing guarantees coordination to use.

**3.1.** The Learning Process. In order to model the learning process, we (implicitly) assume that a trader does not only take into account his own experiences (as he would in e.g. a reinforcement learning model). He observes the prices and the quantities of both platforms (including the observation of the inactiveness of a platform), and takes this information into account. We also assume that an individual trader does not have enough information on other traders or is not able to perform all the necessary computations in order to predict the future behavior of the other traders. Hence, an individual trader cannot accurately predict the future outcomes of the platforms. Furthermore, traders also lack the capability necessary to always compute an exact (but myopic) best reply to the current choices of all other traders.

What can a trader do in such a situation? From his individual, myopic standpoint, if he considers himself to be small relative to market size, the best thing he can do is to evaluate the outcomes of both platforms, and switch to the other platform if he perceives the other platform's outcome as better. A trader can perceive this behavior as approximately rational, since when he switches, the implied changes in prices and traded quantities will most of the time be small, and hence this behavior is close to best reply. Of course, this could also be interpreted as imitation of successful traders of the own market type. We want to stress, though, that the described behavior does not require the observation of any evaluation conducted by other traders, but merely the observation of prices and traded quantities of both platforms.

We proceed now to model the learning process. The state space is given by  $\Omega = \{1,2\}^{|N|} \times \{1,2\}^{|M|}$ . A typical state  $\omega$  specifies which trading platform is chosen by each buyer and each seller - trader k's platform choice is denoted by  $\omega(k) \in \{1,2\}$ . The following notation will prove convenient:

$$N_i(\omega) = \{n \in N | \omega(n) = i\}$$
  
$$M_i(\omega) = \{m \in M | \omega(m) = i\}$$

i.e.  $N_i(\omega) \subseteq N$  is the set of buyers who are on platform *i* in state  $\omega$ , and  $M_i(\omega) \subseteq M$ the set of sellers who are on platform *i* in state  $\omega$ . Of course, for  $j \neq i$ ,  $N_i(\omega) = N \setminus N_j(\omega)$  and  $M_i(\omega) = M \setminus M_j(\omega)$ .

The state of the process at time t = 0, 1, 2, ... is given by  $\omega(t) \in \Omega$ . That is,  $\omega(t)(k) \in \{1, 2\}$  denotes the platform chosen by trader k at time t.

Unperturbed Learning Process. We first concentrate on the unperturbed learning process, where traders switch platform only because of learning, but not because of experimentation (experimentation is introduced in section 3.1 below). If an agent is able to revise his choice for a given period t + 1, he takes the new market outcomes of both platforms in period t and evaluates them. As explained above, we postulate the following learning rule:

**Assumption A:** A trader, who gets the opportunity to revise, observes the outcomes of both platforms in the last period. He then chooses the platform whose outcome he evaluates as best. In case of indifference, he randomizes the choice, with both platforms chosen with strictly positive probability. Choice probabilities may depend on the outcomes but not on the platforms' names.<sup>11</sup>

That is, provided that trader k receives revision opportunity at period t, he will choose the platform with a period t - 1 outcome that he evaluates highest. If, by chance, the outcomes of both platforms are equally evaluated, players are indifferent and will select one platform at random. For instance, in the case in which one platform is inactive and the other is active but yields exactly zero profits for the sellers, sellers randomize among them.

But when are agents allowed to revise their choices? It is common in learning models to explicitly introduce some inertia allowing for the possibility that not all agents

<sup>&</sup>lt;sup>11</sup>In other words, we assume platform symmetry, i.e. the trader's choice does not depend on the identity of the platform but on its observable features - prices and quantities. In case of indifference, both choice probabilities have to be strictly positive but might still depend on other observable characteristics as e.g. the number of traders at each platform.

are able to revise strategies simultaneously (or, for instance, accounting for idiosyncratic switching costs). Different specifications of how revision opportunities arrive give rise to different dynamics and often affect the results. Rather than adopting a specific formulation, here we follow Alós-Ferrer and Kirchsteiger [2] and postulate a general class of dynamics encompassing the standard examples (and many others), which are then reviewed below. See Alós-Ferrer [1] for a discussion.

Let  $E(k, \omega)$  denote the event that agent k receives revision opportunity when the state is  $\omega$ , and let  $E^*(k, \omega) \subseteq E(k, \omega)$  denote the event that agent k is the only agent of his type (i.e. the only buyer or the only seller) receiving revision opportunity in  $\omega$ .

## Assumption B1: $\Pr(E^*(k, \omega)) > 0$ for every agent k and state $\omega$ .

Assumption B1 implies  $\Pr(E(k, \omega)) > 0$ , i.e. every agent has strictly positive probability of being able to revise at any given state. Further, we introduce a weak form of independence between revision opportunities in the two different populations of agents (it can be thought of as an anonymity requirement).

**Assumption B2:** For every agent k and state  $\omega$ , either  $\Pr(E^*(k,\omega) \cap E^*(k',\omega)) > 0$  for any agent k' of the other type, or  $\Pr(E^*(k,\omega) \cap E(k',\omega)) = 0$  for any such k'.

This assumption explicitly excludes non-anonymous situations where, say, whenever seller number 17 gets the opportunity to revise, buyers 3 and 6 also get the opportunity to revise. Assumptions B1 and B2 are rather general. They are fulfilled by the standard models considered in the literature of learning in games. In these models, revision opportunities are either modelled through independent probabilities (a case we call independent inertia; see e.g. Samuelson [38] or Kandori and Rob [27]) or assumed to arrive in an asynchronous way (also called non-simultaneous learning; see e.g. Blume [8], Binmore and Samuelson [7] or Benaïm and Weibull [6]).<sup>12</sup> That is, our formulation covers the following standard examples. Part of our later results will refer to some of these examples.

**Independent Inertia.** There is an exogenous, independent (across traders and periods) probability  $0 < 1 - \rho < 1$  such that the agent does not get revision opportunity in a given state (inertia). Obviously,  $\Pr(E^*(n,\omega)) = \rho(1-\rho)^{n-1} > 0$  for any buyer n, and analogously for sellers, hence verifying B1. B2 follows from independence:  $\Pr(E^*(n,\omega) \cap E^*(m,\omega)) = \Pr(E^*(n,\omega)) \cdot \Pr(E^*(m,\omega)) > 0$  for any buyer n and any seller m.

**Independent Inertia within Types.** In our case, it would be natural to postulate different inertia parameters for buyers and sellers,  $\rho_B, \rho_S \in [0, 1[$ . Clearly, B1 and B2 are not affected by this modification. Notice that, as long as  $\rho_B = \rho_S$  is not excluded, this case encompasses the previous one.

<sup>&</sup>lt;sup>12</sup>The reason we explicitly choose Assumptions B1, B2 is that, in the literature of learning in games, many models are not robust to minute changes in the dynamic assumptions. We want to make explicit that our model is not so sensible to the details of the dynamics.

Asynchronous Learning. Each period, only one agent (i.e. either a buyer or a seller) is (randomly) selected and allowed to revise his strategy. Hence,  $\Pr(E^*(k,\omega)) = \frac{1}{|N|+|M|}$  for any trader k (verifying B1), and  $\Pr(E^*(n,\omega) \cap E(m,\omega)) = 0$  for any pair of buyers and sellers (verifying B2).

Asynchronous Learning within Types. Again, in our case, it is natural to differentiate buyers and sellers and conceive a dynamics where in every period, only one buyer and one seller are selected (randomly and independently) and given the opportunity to revise. Assumption B1 holds because  $\Pr(E^*(n,\omega)) = \frac{1}{|N|} > 0$  for any buyer n and  $\Pr(E^*(m,\omega)) = \frac{1}{|M|} > 0$  for any seller m. Assumption B2 holds by independence.

Note that the second part of Assumption B2 is the one that specifically allows for dynamics where only one agent at all is allowed to revise each period (asynchronous learning). If this part were dropped (which still allows for asynchronous learning within types and independent inertia), the modified Assumption B2 would imply B1.

The specification above allows for more general learning processes than those described by independent inertia or asynchronous learning. Since the revision probability  $\Pr(E(k,\omega))$  is a function of the state  $\omega$ , it might depend e.g. on the difference between the evaluation of the outcomes of both platforms (so that unsatisfied traders are more likely to revise), or on idiosyncratic characteristics of the currently chosen platform.

Assumptions A, B1, and B2 define a stationary Markov chain<sup>13</sup> on the (finite) state space  $\Omega$ . Given two states  $\omega, \omega' \in \Omega$ , denote by  $P^0(\omega, \omega')$  the probability of transition from  $\omega$  to  $\omega'$  in one period for the unperturbed learning process, i.e. for the process without experimentation. The transition matrix of the process is given by  $P^0 = [P^0(\omega, \omega')]_{\omega, \omega' \in \Omega}$ . In the following Lemma we characterize the unperturbed learning process by its absorbing states (states that once entered are never abandoned).

Lemma 1. Assume A, B1, and B2.

- (a) If  $\beta_i > 1$  and  $\beta_j > 1$ , the only absorbing sets of the unperturbed dynamics are the two singletons made of monomorphic states  $\omega_i^*$  (i = 1, 2) such that  $N_i(\omega_i^*) = N, M_i(\omega_i^*) = M.$
- (b) If  $\beta_i > 1$  and  $\beta_j \leq 1$ , the unperturbed dynamics is ergodic and the only absorbing set is the singleton made of the monomorphic state  $\omega_i^*$ .
- (c) If  $\beta_i \leq 1$  and  $\beta_j \leq 1$ , the unperturbed dynamics is irreducible, i.e. the whole state space  $\Omega$  forms an absorbing set.

The previous Lemma shows that depending on the properties of the platforms the unperturbed process is ergodic (the respective absorbing set is unique, see (b) and

 $<sup>^{13}</sup>$ In Appendix C we define and explain all the terms related to the Markov processes of this and the following sections.

(c)) or there is more than one absorbing set (see (a)). In order to select among them, we study in the next section the stability properties of the platforms with respect to experimentation.

**Perturbed Learning Process.** Following the literature on this type of learning models, we proceed to study stochastic stability with respect to experimentation. To do so, the dynamics is enriched with a perturbation in the form of experiments (or mistakes) as follows. With an independent, small probability  $\varepsilon > 0$ , each agent, in each round, might experiment (or make a mistake or "mutate"), and simply pick a platform at random,<sup>14</sup> independently of other considerations.

The dynamics with experimentation is called *perturbed* learning process. Its transition matrix is denoted by  $P^{\varepsilon}$ . Given two absorbing sets X and Y, let  $\gamma(X, Y) > 0$ (referred to as the *transition cost* from X to Y) denote the minimal number of experiments necessary for a direct transition from X to Y, i.e. a positive probability path starting in an element of X and leading to an element in Y, which does not go through any other absorbing set.

Since experiments make transitions between any two states possible, the perturbed process has a single absorbing set formed by the whole state space (i.e. the process is irreducible). The following theorem characterizes the perturbed learning process by its stochastically stable states.<sup>15</sup> In the sequel, whenever we say absorbing sets or states, we refer to the unperturbed dynamics. Since the perturbed dynamics is irreducible, no confusion should arise.

In our case, the stochastic stability analysis is simple because of Lemma 1.

**Theorem 2.** Assume A, B1, and B2.

- (a) If  $\beta_i > 1$  and  $\beta_j > 1$ , only the two monomorphic states can be stochastically stable. Further,  $\omega_i^*$  is stochastically stable if and only if  $\gamma(\omega_i^*, \omega_i^*) \ge \gamma(\omega_i^*, \omega_i^*)$ .
- (b) If  $\beta_i > 1$  and  $\beta_j \leq 1$ , the only stochastically stable state is the monomorphic state  $\omega_i^*$ .
- (c) If  $\beta_i \leq 1$  and  $\beta_j \leq 1$ , all states in  $\Omega$  are stochastically stable.

**Proof.** (b) and (c) are trivial, since the set of stochastically stable states is always non-empty, only states in absorbing sets can be stochastically stable, and all states in the same absorbing set are simultaneously either stochastically stable or not. Part (a) follows from standard results; see the last point of Appendix C.  $\blacksquare$ 

<sup>&</sup>lt;sup>14</sup>We mean that an institution is picked up according to a pre-specified probability distribution having full support, for instance uniformly. The exact distribution does not affect the results, as long as it has full support, and does not depend on  $\varepsilon$ .

<sup>&</sup>lt;sup>15</sup>An absorbing state of the unperturbed process is called stochastically stable if it is in the support of its limit invariant distribution, i.e. if there is a strictly positive probability that the process is in the absorbing state in the long-run and in the limit  $\varepsilon \to 0$ . For a more detailed explanation, formal definitions, and basic characterization results see Appendix C.

**3.2. The long run trading patterns.** We now proceed to analyze the stochastic stability of platforms depending on the values of the parameters  $s_i = (\beta_i, f_i)$  and  $s_j = (\beta_j, f_j)$ .

As a benchmark, we start with the case of identical platform design. To analyze platforms with identical characteristics  $(s_i = s_j)$ , we observe that, for every state  $\omega \in \Omega$  we can uniquely define a so-called *mirror state*  $\widetilde{\omega}$  by changing the platform affiliation of all traders, that is,  $\widetilde{\omega}$  is the only state such that  $M_j(\widetilde{\omega}) = M_i(\omega)$  and  $N_j(\widetilde{\omega}) = N_i(\omega)$ . Then,

**Lemma 3.** Suppose  $s_i = s_j$ . Then,  $\mu^*(\omega) = \mu^*(\widetilde{\omega}) \ \forall \omega \in \Omega$ .

**Proof.** Follows directly from  $P(\omega, \omega') = P(\widetilde{\omega}, \widetilde{\omega}') \quad \forall \omega, \omega' \in \Omega$  which holds for  $s_i = s_j$  due to platform symmetry (recall Assumption A).

Theorem 2 already identifies the set of stochastically stable states whenever at least one platform *i* has a price bias  $\beta_i \leq 1$ . Hence we are left with design configurations  $s_i$  and  $s_j$  where both price biases favor sellers (i.e.  $\beta_i, \beta_j > 1$ ). There, Theorem 2 indicates that full coordination on both platforms are the only possible stochastically stable states. To further pin down the long-run trading pattern, it proves useful to distinguish the two platforms with respect to their prices. We start with the platform which implements a (weakly) lower price.

**Lemma 4.** Suppose  $\beta_i, \beta_j > 1$  and  $p_i = \frac{\beta_i c}{1-f_i} \leq \frac{\beta_j c}{1-f_j} = p_j$ . Then,  $\gamma(\omega_i^*, \omega_j^*) \geq 2 = \gamma(\omega_j^*, \omega_i^*)$  and  $\omega_i^*$  is stochastically stable.

**Proof.** If  $p_i \leq p_j$ , buyers (weakly) prefer platform i to platform j whenever it is active. Accordingly,  $\omega_i^*$  can be reached from  $\omega_j^*$  with just two mutations. To see this, suppose the system is in state  $\omega_j^*$ . A buyer and a seller switching to platform idue to experimentation induce trade at platform i. B1 and B2 guarantee that there is a positive probability that first all buyers at platform j get the opportunity to revise their platform choice (and switch to i because of the lower price or because of indifference) and subsequently all sellers at platform j can revise and also switch to platform i as this is now the only platform that generates positive profits for them. Hence,  $\gamma(\omega_j^*, \omega_i^*) = 2$ . Clearly  $\gamma(\omega_i^*, \omega_j^*) \geq 2$  as at least one seller and one buyer have to switch to platform j to induce trade. Then, Theorem 2(a) implies stochastic stability of  $\omega_i^*$ .

We now turn to the stochastic stability of a platform with a strictly higher price (provided that the biases of both platforms favor sellers). While all our previous results did not depend on the modelling details such as (i) absolute population size of buyers and sellers, (ii) the relative size of these populations, (iii) the heterogeneity of buyers, (iv) the price elasticity of demand, (v) the grid size  $\delta$ , and (vi) details of the learning process (e.g. adjustment speed, asymmetries between buyers and sellers), these details do matter now as the following results illustrate.

**Lemma 5.** Suppose  $\beta_i, \beta_j > 1$ ,  $p_i = \frac{\beta_i c}{1-f_i} < \frac{\beta_j c}{1-f_j} = p_j$ , so that  $\omega_i^*$  is stochastically stable.

(a) in a dynamics with independent inertia,  $\omega_j^*$  is also stochastically stable (i.e.  $\gamma(\omega_i^*, \omega_j^*) = 2$ ) if and only if there is at least one buyer  $\tilde{n} \in N$  such that

$$d_{\tilde{n}}\left(\frac{\beta_j c}{1-f_j}\right)(\beta_j-1) \ge \frac{1}{|M|-1} D_{N \setminus \{\tilde{n}\}}\left(\frac{\beta_i c}{1-f_i}\right)(\beta_i-1).$$

(b) in a dynamics with asynchronous learning,  $\omega_j^*$  is also stochastically stable (i.e.  $\gamma(\omega_i^*, \omega_j^*) = 2$ ) if and only if there is at least one buyer  $\tilde{n} \in N$  such that

$$\frac{1}{|M|-1}d_{\tilde{n}}\left(\frac{\beta_j c}{1-f_j}\right)(\beta_j-1) \ge D_{N\setminus\{\tilde{n}\}}\left(\frac{\beta_i c}{1-f_i}\right)(\beta_i-1).$$

Note that the condition in Lemma 5(a) is violated whenever e.g. buyers are identical,  $|N| \ge |M|$ , and the price elasticity of demand is sufficiently high. In contrast, the condition can be satisfied for  $\beta_j > \beta_i$  whenever e.g. buyers are sufficiently heterogeneous (i.e.  $\exists \ \tilde{n} \in N$  such that  $d_{\tilde{n}}(p) >> d_n(p) \ \forall n \neq \tilde{n}$ ), or buyers are identical and |M| >> |N|, or d(p) is sufficiently inelastic.

The stochastic stability of  $\omega_j^*$  is harder to establish if the dynamics is slow as e.g. under asynchronous learning. The condition in Lemma 5(b) is violated whenever e.g. buyers are identical and the price elasticity of demand is sufficiently high (in contrast to the case of independent inertia, this holds independently of the sizes of populations |M| and |N|). The condition can be fulfilled for  $\beta_j > \beta_i$  if e.g. buyers are sufficiently heterogeneous or buyers are identical and demand is sufficiently inelastic.

**Remark 1.** The analysis in the previous lemmata establishes stochastic stability through transition paths involving at most two simultaneous mutations. In the terminology of Ellison [13], this implies that the estimated time of first arrival, i.e. the estimated time until a stochastically stable state is first observed, is of order  $\varepsilon^{-2}$ . Thus the speed of convergence is relatively high. The number of mutations needed for the relevant transitions does not increase with the population size. Hence, our dynamics escape the well-known critique that for large populations the long run may actually be "too long" to be relevant (see Kandori et al. [26] or Ellison [12]).

**3.3.** Platform Revenues and Designers' Profits. Till now we have analyzed the learning dynamics of the traders and the resulting long run pattern of trades. Next turn to the revenues generated by the platforms, which in turn determine the profits of the market designers.

When analyzing the market designers' choice of the characteristics of the trading platforms we will assume that platform designers are long-lived, patient, and (relatively) rational agents when compared with individual buyers or sellers. Hence, the designers take only the long term expected revenues into account, and ignore revenues made during the adjustment process to the limit invariant distribution.<sup>16</sup> Given the platform characteristics  $s = (s_i, s_j)$ , the long-run expected revenues per round  $ER_i(s)$  depend on the limit invariant distribution. The profits of designer *i* are given by  $\pi_{D,i}(s) = f_i ER_i(s)$  implying that  $\pi_{D,i}(s) \ge 0$  for all *s*.

Consider first a platform i with  $\beta_i < 1$ .

**Lemma 6.** Suppose  $\beta_i < 1$ . Then  $\pi_{D,i}((\beta_i, f_i), s_j) = 0$  for all feasible  $f_i, s_j$ .

**Proof.** Follows immediately form the fact that trade is never possible on platforms with  $\beta_i < 1$ .

Hence, we are left with platform configurations  $(s_i, s_j)$  where both platforms have a price bias weakly larger than one. In this case expected revenues at platform *i* depend not only on the design of this platform but also on the design of the other platform as the following results indicate.

**Lemma 7.** Consider a platform configuration  $s = (s_i, s_j)$  with  $s_i = (\beta_i, f_i)$ ,  $s_j = (\beta_j, f_j)$  and prices  $p_i = \beta_i \frac{c}{1-f_i}$ ,  $p_j = \beta_j \frac{c}{1-f_j}$ .

- (a) If  $s_i = s_j$ , then,  $\pi_{D,k}(s) = \frac{1}{2} f_k p_k D_N(p_k) > 0$  for k = 1, 2.
- (b) If  $\beta_i = \beta_j = 1$  and  $f_i < f_j$ , then  $f_k p_k D_N(p_k) > \pi_{D,k}(s_i, s_j) > 0$  for k = 1, 2.
- (c) If  $\beta_i > 1$  and  $\beta_j \leq 1$ , then  $\pi_{D,i}(s_i, s_j) = f_i p_i D_N(p_i)$  and  $\pi_{D,j}(s_j, s_i) = 0$ .
- (d) If  $\beta_i, \beta_j > 1$ ,  $p_i \leq p_j$ , and  $\gamma(\omega_i^*, \omega_j^*) > 2$ , then  $\pi_{D,i}(s_i, s_j) = f_i p_i D_N(p_i)$  and  $\pi_{D,j}(s_i, s_j) = 0$ .
- (e) If  $\beta_i, \beta_j > 1, p_i \le p_j$ , and  $\gamma(\omega_i^*, \omega_j^*) = 2$ , then  $\pi_{D,k}(s_i, s_j) = \mu^*(\omega_k^*) f_k p_k D_N(p_k) > 0$  for k = 1, 2.

**Proof.** (a) follows directly from Lemma 3 as the price and the traded quantity in  $\omega$  at platform *i* are identical to price and quantity in  $\tilde{\omega}$  at platform *j*. To see (b), recall from Theorem 2(c) that every  $\omega \in \Omega$  is stochastically stable if  $\beta_i = \beta_j = 1$ . Hence, there is a strictly positive probability for trade (i.e. positive revenues) at each platform. (c) follows directly from Theorem 2(b). (d) and (e) follow directly from Theorem 2(a) and Lemma 4.

### 4. The Platform Design

We now compare the design choices by a monopolistic designer and by two competing designers. Observe for reference that, as shown in the last Lemma 7, the designers' expected revenues are always weakly positive.

<sup>&</sup>lt;sup>16</sup>Otherwise, the payoffs of a market designer would not only depend on the characteristics of both platforms, but also on the initial distribution of the traders over the platforms. In the absence of a plausible theory on the initial distribution, the results would be arbitrary. Further, as pointed out in Remark 1, convergence to full coordination is fast, and hence the assumption is, to some extent, justified.

4.1. Monopolistic Market Design. As a benchmark, we briefly consider the case where only one platform is available, with characteristics  $s = (\beta, f)$ . In this case traders have no choice but to use this platform. Designer's profits are given by:

$$\pi_D(s) = \begin{cases} f \frac{\beta c}{1-f} D_N\left(\frac{\beta c}{1-f}\right) & \text{if } \beta \ge 1\\ 0 & \text{otherwise.} \end{cases}$$

What is the profit maximizing platform  $(\beta^*, f^*)$ ? Note first that for  $\beta < 1$  the profits are zero, whereas for  $\beta \ge 1$  and for 0 < f < 1 the profits are strictly positive. Hence,  $\beta^* \ge 1$ . Now assume for a moment that  $\beta$  and f are continuous variables with  $f \in [0, 1]$  and  $\beta \in [1, \infty)$ . Denote  $p = \frac{\beta c}{1-f}$  and recall that  $\lim_{p\to\infty} d_n(p)p = 0$  for all  $n \in N$ . Hence, it must hold that  $0 < f^* < 1$ .

Differentiating the designer's profits yields (for  $\beta \geq 1$ )

$$\frac{\partial \pi_D}{\partial f}(\beta, f) = pD_N(p) + f\frac{\partial p}{\partial f}[D_N(p) + pD'_N(p)]$$
$$\frac{\partial \pi_D}{\partial \beta}(\beta, f) = f\frac{\partial p}{\partial \beta}[D_N(p) + pD'_N(p)]$$

where  $\frac{\partial p}{\partial f} = \frac{\beta c}{(1-f)^2} > 0$  and  $\frac{\partial p}{\partial \beta} = \frac{c}{1-f} > 0$ . Let the optimal price be  $p^* = \frac{\beta^* c}{1-f^*}$ . Since  $0 < f^* < 1$ , the first order conditions for the designer's optimum imply that  $\frac{\partial \pi_D}{\partial f}(\beta^*, f^*) = 0$ , thus  $D_N(p^*) + pD'_N(p^*) < 0$ . This implies that  $\frac{\partial \pi_D}{\partial \beta}(\beta^*, f^*) < 0$ , hence the designer's profits are maximized at the corner solution  $\beta^* = 1$ .

Of course, in our model  $\beta$  and f are not continuous variables. However, if the grid of feasible fees is fine enough, the optimal fee approximates the one of the continuous case, and hence the optimal  $\beta$  is 1 also in the discontinuous case. Hence we conclude that a monopolistic market designer would introduce a market clearing platform, because such an platform allows him to reap the highest profits.

The intuitive reason for this result is as follows. Suppose revenues  $pD_N(p)$  are maximized at price  $p^*$ . Note that this price can be attained with different  $(\beta, f)$  combinations and that  $p^* = \frac{\beta c}{1-f}$  is increasing both in  $\beta$  and f. Since the monopolist designer's profits are  $fpD_N(p)$ , he will try to reach  $p^*$  with that  $(\beta, f)$  combination that has the highest fee, and hence the lowest possible  $\beta \geq 1$ .

4.2. Competitive Market Design. In order to reflect that platform designers are "more rational" than individual buyers and sellers, we simply consider them rational players in the normal-form game defined by these payoff functions.<sup>17</sup> That is, both designers choose their platforms simultaneously and payoffs are given as above. The sets of pure strategies of designer *i* and *j* are given by  $S_i = S_j = B \times F$ . We also allow designers to use mixed strategies, i.e. choose a probability distribution over *S* rather than picking up a particular characteristic for sure.

<sup>&</sup>lt;sup>17</sup>Interestingly enough, it is possible to build a model with boundedly rational market designers where or main results still hold. See Appendix B.1.

ON THE EVOLUTION OF MARKET INSTITUTIONS: THE PLATFORM DESIGN PARADOX 18

Denote by  $\sigma_i$  the (mixed) strategy of designer *i*. The expected payoff of *i* is

$$\pi_{D,i}(\sigma_i,\sigma_j) = \sum_{s_j \in S} \sum_{s_i \in S} \sigma_j(s_j) \sigma_i(s_i) f_i E R_i(s_i,s_j).$$

Further,  $\pi_{D,i}(s_i, \sigma_j)$  denotes the expected payoff if designer *i* chooses  $s_i$  for sure and *j* chooses the probability distribution  $\sigma_j$ .

Since the sets of pure strategies are finite, a Nash equilibrium of the designers' game always exists (possibly in mixed strategies). To characterize these equilibria, we need the following Lemma.

**Lemma 8.** Let  $(\sigma_i^*, \sigma_j^*)$  be a Nash equilibrium (possibly in mixed strategies). Then, for any pure strategy  $s_i = (\beta_i, f_i)$  of player *i* such that  $\sigma_i^*(s_i) > 0$ , it holds that  $\beta_i \ge 1$ .

**Proof.** Assume to the contrary that there exists a pure strategy  $\overline{s}_i = (\overline{\beta}_i, \overline{f}_i)$  with  $\sigma_i^*(\overline{s}_i) > 0$  and  $\overline{\beta}_i < 1$ . By Lemma 6 this pure strategy gives designer *i* a profit of zero against all strategies of *j*. Hence,  $\pi_{D,i}(\overline{s}_i, \sigma_j^*) = 0$ , and, since  $\sigma_i^*$  is an equilibrium strategy,  $\pi_{D,i}(\sigma_i^*, \sigma_j^*) = 0$ .

Suppose that, in equilibrium, j chooses only platforms with  $\beta_j < 1$ . That is,  $\beta_j < 1$  for all  $s_j = (\beta_j, f_j) \in S$  with  $\sigma_j^*(s_j) > 0$ . In this case, if designer i chooses with certainty a platform  $s'_i$  with  $\beta'_i > 1$ , Lemma 7(c) implies that  $\pi_{D,i}(s'_i, \sigma^*_j) =$  $f'_i \frac{\beta'_i c}{1-f'_i} D_N\left(\frac{\beta'_i c}{1-f'_i}\right) > 0$ . Since  $\pi_{D,i}(\sigma^*_i, \sigma^*_j) = 0$ , this contradicts that  $(\sigma^*_i, \sigma^*_j)$  is a Nash equilibrium.

Thus, there must exist an  $\overline{s}_j$  with  $\overline{\beta}_j \geq 1$  such that  $\sigma_j^*(\overline{s}_j) > 0$ . Then, if designer *i* deviates to the pure strategy  $s'_i = \overline{s}_j$ ,  $\pi_{D,i}(s'_i, \sigma^*_j) = \sigma^*_j(\overline{s}_j)f_iER_i(s'_i, \overline{s}_j) + \sum_{\substack{s_j \in S \setminus \overline{s}_j \\ \pi_{D,i}(s'_i, \sigma^*_j) > 0} \sigma_j^*(\overline{s}_j)\pi_{D,i}(s'_i, s_j)$ . Since by Lemma 7(a)  $ER_i(s'_i, \overline{s}_j) > 0$ , we conclude that  $\pi_{D,i}(s'_i, \sigma^*_j) > 0$ , again contradicting that  $(\sigma^*_i, \sigma^*_j)$  is a Nash equilibrium.

Hence, we have shown that, in equilibrium, only platforms weakly biased in favor of the sellers can be chosen. We now want to show that, actually, in any equilibrium, both designers will introduce platforms that lead to prices strictly above the market clearing level - platforms that lead to market clearing prices will not be designed in equilibrium. This results holds as long as the grid of possible biases is fine enough.

**Theorem 9.** Let  $(\sigma_i^*, \sigma_j^*)$  be a Nash equilibrium (possibly in mixed strategies). For any pure strategy  $s_i = (\beta_i, f_i)$  of player *i* such that  $\sigma_i^*(s_i) > 0$ , it holds that  $\beta_i > 1$  if  $\delta$  is sufficiently small.

**Proof.** By the previous lemma, only platforms with  $\beta \geq 1$  will be designed in equilibrium. Assume by contradiction that there exist some pure strategies  $s_i = (\beta_i, f_i)$  with  $\sigma_i^*(s_i) > 0$  and  $\beta_i = 1$ . Denote a strategy of this type by  $\overline{s}_i = (1, \overline{f}_i)$  and let  $\overline{p} = \frac{c}{1-\overline{f}_i}$ . Denote the carrier or support of  $\sigma_j^*$  by  $C(\sigma_j^*) = \{s_j = (\beta_j, f_j) \in S | \sigma_j^*(s_j) > 0\}$ .

Suppose that, for all  $s_j \in C(\sigma_j^*)$  we actually had that  $\beta_j > 1$ . This implies by Lemma 7(c) that  $\pi_{D,i}(\sigma_i^*, \sigma_j^*) = \pi_{D,i}(\overline{s}_i, \sigma_j^*) = 0$ . Take any  $s'_i = s'_j \in C(\sigma_j^*)$ . By Lemma 7(a), and recalling that  $\pi_{D,i}(s_i, s_j) \ge 0$  for all  $s_i, s_j$ , we obtain that

$$\pi_{D,i}(s'_i, \sigma^*_j) \ge \sigma^*_j(s'_j) f'_j E R_j(s'_i, s'_j) > 0.$$

Hence, player i would have an incentive to deviate from  $\sigma_i^*$ , a contradiction.

We conclude that there exists some  $s_j \in C(\sigma_j^*)$  with  $\beta_j = 1$ . Let  $C_1(\sigma_j^*) = \{s_j = (\beta_j, f_j) \in C(\sigma_j^*) | \beta_j = 1\}$ . Notice that, since  $\overline{s}_i = (1, \overline{f}_i)$ , we have by Lemma 7(c) that  $\pi_{D,i}(\overline{s}_i, s_j) = 0$  for all  $s_j \in C(\sigma_j^*)$  with  $\beta_j > 1$ . Then, by Lemma 7(a,b),

$$\pi_{D,i}(\overline{s}_i,\sigma_j^*) < \sum \left\{ \sigma_j^*(s_j) \overline{f_i} \overline{p} D_N(\overline{p}) \, \big| s_j \in C_1(\sigma_j^*) \right\}$$

However, for any  $s'_i$  with  $\beta'_i > 1$  and  $f'_i = \overline{f_i}$ ,

$$\pi_{D,i}(s'_i,\sigma^*_j) \ge \sum \left\{ \sigma^*_j(s_j) \overline{f_i} \beta'_i \overline{p} D_N\left(\beta'_i \overline{p}\right) \middle| s_j \in C_1(\sigma^*_j) \right\}$$

due to Lemma 7 (c) (the inequality follows from the fact that  $\pi_{D,i}(s'_i, s_j) \geq 0$  for all  $s_j$ ). This latter expression is continuous in  $\beta'_i$ . Thus, for  $\beta'_i$  approaching one from above,  $\pi_{D,i}(s'_i, \sigma^*_j) > \pi_{D,i}(\overline{s}_i, \sigma^*_j) = \pi_{D,i}(\sigma^*_i, \sigma^*_j)$ . Hence, if the grid is fine enough<sup>18</sup> player *i* has an incentive to deviate from  $\sigma^*_i$  to an institution with  $\beta'_i > 1$  but close enough to 1. A contradiction.

We have thus established the paradoxical result that competition among platform designers will induce them to select biased platforms which implement noncompetitive market outcomes. In general, nothing more can be said about the specific characteristics of the Nash equilibria. A brief examination of Lemma 5 and Lemma 7(d) and (e) should convince the reader that a full characterization of the Nash equilibria will depend on the exact shape of the limit invariant distribution, and not only on its support. This distribution in turn depends on the details of the dynamics, e.g. whether learning opportunities arise simultaneously among traders or asynchronously. In contrast, the last theorem holds for any specification of the learning dynamics satisfying assumptions B1 and B2.

Still, one might suspect that competition leads to platforms close to the market clearing one, i.e. to platforms with  $\beta_i = 1 + \delta$ . If this hypothesis would be correct, the chosen platforms would nearly resemble market clearing ones as long as the grid of feasible biases is fine enough. The next proposition, however, shows that this hypothesis is in general false. For simplicity, consider identical buyers with a demand function d(p) and denote the price elasticity of demand by  $\varepsilon_p(p) = -\frac{pd'(p)}{d(p)}$ .

**Proposition 10.** Assume independent inertia, identical buyers and |M| = |N|. If the grids B and F are fine enough and  $\varepsilon_p$  is not much larger than one, then

 $<sup>^{18}</sup>$ The grid can be assumed to be *ex ante* fine enough by a uniform continuity argument.

i) there exists no Nash equilibrium  $(\sigma_i^*, \sigma_j^*)$  (possibly in mixed strategies) where both designers introduce only platforms with  $\beta_i = \beta_j = 1 + \delta$ ;

ii) there exists no pure strategy Nash-equilibrium  $(s_i^*, s_j^*)$  of the design game with  $\beta_i^* = \beta_j^* = 1 + \delta$ .

This proposition shows that equilibrium designs (beyond the features highlighted in Theorem 9) are in general rather sensitive to details of the economy and the learning process. For example it cannot be excluded that in equilibrium designers choose "near market clearing" platform characteristics for some specifications of the learning dynamics and/or of the demand functions. But in general the equilibrium choices are not "near market clearing" platforms.

#### 5. DISCUSSION

We have shown that if several trading platforms are available, traders will learn to coordinate on a platform with prices systematically above the market clearing level, if such a platform is feasible. This forces competing market designers to create such non-market clearing platforms. On the other hand a monopolistic market designer will always introduce a market clearing platform in order to maximize his profits. Hence, we derive the paradoxical result that platform competition induces non-competitive market outcomes. This result could also explain why so many B2B platforms exhibit institutional designs that are notorious for biased (non-market clearing) prices (e.g. posted offer markets, proxy auctions with "hard-close" or Dutch auctions).

The result of our paper depends of course on several assumptions we have made. First, we have assumed sellers to be producers endowed with a technology with constant returns to scale. Although this is a focal, economically meaningful case, it clearly simplifies the analysis and allows for a clear-cut derivation of the results. Under production technologies exhibiting decreasing returns to scale, the results are not so sharp and a characterization of the limit invariant distribution requires both a further specification of the learning behavior of the traders and a further specification of demand and supply. In Appendix B we provide an extended example with decreasing returns to scale where our main result still holds. It shows, however, that the optimality of a price bias is no longer independent of details like learning velocities. Nonetheless, this clearly illustrates that the scope of the paradox identified here goes beyond the constant returns to scale case.

Second, we have focused on an asymmetric rationality model where market designers are sophisticated when compared to traders. Furthermore, by focusing on long-run profits we have implicitly assumed that it is much more difficult for designers to change the properties of their platforms than for traders to switch trading platforms. In Appendix B we analyze the case of boundedly rational designers who have to learn how to design a platform through a regular (trial-and-error) design revision process. Our main result - the emergence of non-market clearing institutions carries over to such a setting.

These robustness checks show that neither the assumption of constant returns to scale technology nor that of rational designers *drive* our results. Rather, it is indeed

platform competition that leads to the emergence of non-market clearing trading platforms.

#### References

- Alós-Ferrer, C. (2003), Finite Population Dynamics and Mixed Equilibria, International Game Theory Review 5 (3), 263-290.
- [2] Alós-Ferrer, C. and G. Kirchsteiger (2003), Does Learning lead to Coordination in Market Clearing Institutions?, Working Paper 0320, Department of Economics, University of Vienna.
- [3] Ariely, D., A. Ockenfels, and A. Roth (2005), An Experimental Analysis of Ending Rules in Internet Auctions, RAND Journal of Economics, 36, 790-809.
- [4] Armstrong, M. (2002), Competition in Two-sided Markets, mimeo.
- [5] Belleflamme, P. and E. Toulemonde (2004), Emergence and Entry of B2B Marketplaces, mimeo.
- [6] Benaïm. M. and J. Weibull (2003), Deterministic Approximation of Stochastic Evolution in Games, Econometrica 71, 878-903.
- [7] Binmore, K. and L. Samuelson (1997), Muddling Through: Noisy Equilibrium Selection, Journal of Economic Theory 74, 235-265.
- [8] Blume, L. (1995), The Statistical Mechanics of Best-Response Strategy Revision, Games and Economic Behavior 11, 111-145.
- [9] Caillaud, B. and B. Jullien (2002), Chicken and Egg: Competition among Intermediation Service Providers, mimeo.
- [10] Davila, A., M. Gupta, and R. Palmer (2003), Moving Procurement Systems to the Internet: The Adoption and Use of E–Procurement Technology Models, European Management Journal 21 (1), 11-23.
- [11] The Economist May 13th (2004), E-commerce takes off.
- [12] Ellison, G. (1993), Learning, Local Interaction, and Coordination, Econometrica 61, 1047-1071.
- [13] Ellison, G. (2000), Basins of Attraction, Long-Run Stability, and the Speed of Step-by-Step Evolution, Review of Economic Studies 67, 17-45.
- [14] Ellison, G. (2005), Bounded Rationality in Industrial Organization, mimeo.
- [15] Ellison, G., and D. Fudenberg (2003), Knife-edge or Plateau: When Do Markets Models Tip?, Quarterly Journal of Economics 118, 1249-1278.

- [16] Ellison, G., D. Fudenberg, and M. Möbius (2004), Competing Auctions, Journal of the European Economic Association 2, 30-66.
- [17] European Commission, Enterprise Directorate General (2003), Report of the Expert Groupon B2B Internet Trading Platforms, Final report, mimeo.
- [18] Friedman, D. (1993), How Trading Institutions Affect Financial Market Performance: Some Laboratory Evidence, Economic Inquiry 31, 410-435.
- [19] Fudenberg, D., and D. Levine (1998), The Theory of Learning in Games. The MIT Press, Cambridge, Massachusetts.
- [20] Freidlin, M.I. and A.D. Wentzell (1984), Random Perturbations of Dynamical Systems, Springer Verlag, New York.
- [21] Gabaix, X., and D. Laibson (2005), Shrouded Attributes, Consumer Myopia, and Information Suppression in Competitive Markets, forthcoming in Quarterly Journal of Economics.
- [22] Gerber, A., and M.O. Bettzüge (2002), Evolutionary Choice of Markets, forthcoming in Economic Theory.
- [23] Holt, C. (1995), Industrial Organization, in Kagel, J. and A. Roth (eds): Handbook of Experimental Economics, Princeton University Press, Princeton.
- [24] Hopkins, E. (2005), Adaptive Learning Models of Consumer Behavior, mimeo.
- [25] Kagel, J. (1995), Auctions: A Survey of Experimental Research, in: Handbook of Experimental Economics, Princeton.
- [26] Kandori, M., G. Mailath, and R. Rob (1993), Learning, Mutation, and Long-Run Equilibria in Games. Econometrica 61, 29-56.
- [27] Kandori, M., and R. Rob (1995), Evolution of Equilibria in the Long Run: a general theory and applications. Journal of Economic Theory 65, 383-414.
- [28] Kirchsteiger, G., Niederle, M., and J. Potters (2005), Endogenizing Market Institutions: An Experimental Approach, European Economic Review 49(7), 1827-1853.
- [29] Kugler, T., Neeman, Z., and N. Vulkan (2006), Markets versus negotiations: An experimental investigation, Games and Economic Behavior 56, 121-134.
- [30] Lucking-Reiley, D. and Spulber, D. (2001), Business to Business Electronic Commerce, The Journal of Economic Perspectives 15, 55-68.
- [31] Neeman, Z. and N. Vulkan (2000), Markets versus Negotiations: The Predominance of Centralized Markets. mimeo.

- [32] Ockenfels, A. and Roth, A. (2006), Late and Multiple Bidding in Second-Price Internet Auctions: Theory and Evidence Concerning Different Rules for Ending an Auction, Games and Economic Behavior 55, 297-320.
- [33] Plott, C. (1982), Industrial Organization Theory and Experimental Economics, Journal of Economic Literature 20, 1485-1587.
- [34] Plott, C. and V. Smith (1978), An Experimental Examination of Two Exchange Institutions, Review of Economic Studies 45, 133-153.
- [35] Rochet, J. and J. Tirole (2003), Platform Competition in Two-sided Markets, Journal of the European Economic Association 1, 990-1029.
- [36] Rochet, J. and J. Tirole (2004), Two-sided Markets: An Overview, mimeo.
- [37] Roth, A. and A. Ockenfels (2002), Last-Minute Bidding and the Rules for Ending Second-Price Auctions: Evidence from eBay and Amazon on the Internet, American Economic Review 92(4), 1093-1103.
- [38] Samuelson, L. (1994), Stochastic Stability in Games with Alternative Best Replies. Journal of Economic Theory 64, 35-65.
- [39] Samuelson, L. (1997), Evolutionary Games and Equilibrium Selection. The MIT Press, Cambridge, Massachusetts.
- [40] Schlag, K. (2004), Competing for Boundedly Rational Consumers, mimeo.
- [41] Spiegler, R. (2006), Competition over Agents with Boundedly Rational Expectations, Theoretical Economics 1, 207-231.
- [42] UNCTAD (2002), e-Commerce and Development Report 2002, mimeo.
- [43] Young, P. (1993), The Evolution of Conventions. Econometrica 61, 57-84.

## APPENDIX

### A. Proofs

A.1. Proof of Lemma 1. We first prove the following preliminary claim. The monomorphic state  $\omega_i^*$  can be reached with positive probability from any state where platform  $j \neq i$  is inactive. To see this, simply notice that, by Assumptions A and B1, there is positive probability that traders will switch away from the inactive platform j until  $\omega_i^*$  is reached. An immediate consequence of this claim is that, from the "cross-states" where both platforms are inactive, i.e. all buyers are at one platform and all sellers are at the other platform, both monomorphic states can be reached.

(a) Let  $\beta_1 > 1$  and  $\beta_2 > 1$ . First note that the monomorphic states are absorbing, because at the corresponding platform both buyers and sellers make strictly positive profits and the other platform is inactive. Thus traders stay at the active one.

To complete the proof, it is enough to show that there exists a positive probability path from any state to some of the monomorphic states. By the preliminary claim, this is true if any platform is inactive. Consider, thus, an arbitrary  $\omega \in \Omega$  such that  $1 \leq |N_i(\omega)| \leq |N| - 1$  and  $1 \leq |M_i(\omega)| \leq |M| - 1$ .

Without loss of generality, suppose  $p(s_i) \leq p(s_j)$ . Buyers weakly prefer platform i to platform j. By B1 and B2, there is positive probability that all buyers at j receive revision opportunity in successive periods and only sellers at j receive revision opportunity. Hence, buyers will switch away from j and no new seller will switch to j. Thus, either the monomorphic state  $\omega_i^*$  is eventually reached, or j becomes inactive, leading again to  $\omega_i^*$  with positive probability.

(b) Let  $\beta_j \leq 1$  and  $\beta_i > 1$ . Provided *i* is active, sellers strictly prefer it to platform *j*, regardless of prices. As above, the monomorphic state  $\omega_i^*$  is absorbing. The monomorphic state  $\omega_j^*$  is not. To see this, note that sellers always receive zero profits at *j* and will switch to *i* with positive probability even if profits there are zero (the latter due to randomization in case of indifference). Thus, starting at  $\omega_j^*$ , with positive probability *j* becomes inactive such that  $\omega_i^*$  is reached with positive probability.

By the preliminary claim, from any state where platform j is inactive the monomorphic state  $\omega_i^*$  can be reached. If i is inactive, then  $\omega_j^*$  is reached, but then the dynamics leads to  $\omega_i^*$  with positive probability.

Consider an arbitrary  $\omega$  such that  $1 \leq |N_i(\omega)| \leq |N| - 1$  and  $1 \leq |M_i(\omega)| \leq |M| - 1$ . By B1 and B2, there is positive probability that all sellers at j receive revision opportunity in successive periods and only buyers at j receive revision opportunity.

Hence, sellers will switch away from j and no new buyer will switch to j. Eventually, either the monomorphic state  $\omega_i^*$  is reached, or j becomes inactive (which again leads to  $\omega_i^*$  with positive probability).

(c) Last, suppose  $\beta_i \leq 1$  and  $\beta_j \leq 1$ . If both  $\beta_i < 1$  and  $\beta_j < 1$ , both platforms are always inactive. Hence every trader randomizes when given revision opportunity, thus the process is obviously irreducible.

If  $\beta_j < 1$  and  $\beta_i = 1$ , sellers always receive zero profits and always randomize. Buyers, though, switch away from the inactive platform j whenever platform i is active and they receive revision opportunities, while they randomize whenever i is inactive. Thus, from any state, there is always positive probability to reach the monomorphic state  $\omega_i^*$ . From this state, any state in

$$\Omega_i^B = \{ \omega \, | N_i(\omega) = N \}$$

can be reached. Further, since sellers simply randomize, they might eventually all switch to platform j, reaching the cross state in which all sellers are at j and all buyers are at i. As long as all sellers are in j, buyers merely randomize. That is, any

state in

$$\Omega_j^S = \{\omega \, | M_j(\omega) = M \}$$

can be reached. Consider an arbitrary state  $\omega'$  with  $|N_i(\omega')| > 0$ . This state can be reached from a state  $\omega \in \Omega_j^S$  with  $N_i(\omega) = N_i(\omega')$  by letting sellers switch to *i* and giving revision opportunity only to buyers at *i*, who will stay there. Finally, consider a state  $\omega'$  with  $N_i(\omega') = \emptyset$ , i.e.  $N_j(\omega') = N$ . This state can be reached from the monomorphic state  $\omega_j^* \in \Omega_j^S$  by letting sellers randomize appropriately, because buyers who receive revision opportunity are indifferent (both platforms are inactive) and will stay at *j* with positive probability.

In summary, any state can be reached from  $\omega_i^*$ , and the latter can be reached from any state. This completes the argument if  $\beta_j < 1$  and  $\beta_i = 1$ .

The only remaining case is  $\beta_i = \beta_j = 1$ . In this case, sellers always obtain zero profits and hence randomize. Buyers in turn want to move away from inactive platforms. If both platforms are inactive, they randomize, too. However, since sellers simply randomize, it is clear that, from an arbitrary state, any platform might eventually become inactive by lack of sellers. It follows from the preliminary claim that, from any state, both monomorphic states can be reached.

>From the monomorphic states, since sellers still simply randomize, all states at  $\Omega_i^B$  and  $\Omega_j^B$  (see above) can be reached - in particular the cross states. Then, buyers also randomize. As long as sellers stay concentrated at one platform, buyers strictly prefer that platform if it became active. That is, we can construct positive probability paths to any state in  $\Omega_j^S$  and  $\Omega_i^S$  (see above). Again, since sellers merely randomize, we can construct positive probability paths to any state  $\omega'$  with  $0 < |N_i(\omega')| < |N|$  by letting sellers switch appropriately and giving revision opportunity only to buyers who are already at i. This completes the argument if  $\beta_i = \beta_i = 1$ .

**A.2.** Proof of Lemma 5. It has been shown in the proof of Lemma 4 that  $\gamma(\omega_j^*, \omega_i^*) = 2$  if  $\beta_i, \beta_j > 1$  and  $p_i \leq p_j$ . Hence, a necessary and sufficient condition for the stochastic stability of  $\omega_j^*$  is  $\gamma(\omega_i^*, \omega_j^*) = 2$ .

Since  $p_i < p_j$ , buyers never switch to platform j as long as i is active. Hence,  $\omega_j^*$  has to be reached through switching of *all* sellers to platform j and a subsequent switch of all buyers to the only remaining active platform.

In case (a), under independent inertia there is positive probability that all sellers at platform j simultaneously receive the opportunity to revise. If one seller and buyer  $\tilde{n} \in N$  are already present at platform j, sellers will switch to j if  $d_{\tilde{n}}(\frac{\beta_j c}{1-f_j})(\beta_j - 1) \geq \frac{1}{|M|-1}D_{N\setminus\{\tilde{n}\}}\left(\frac{\beta_i c}{1-f_i}\right)(\beta_i - 1)$ . Hence, this condition is sufficient for the stochastic stability of  $\omega_j^*$ . To see that it is also necessary suppose that it is violated. Then no seller will switch to j after one seller and any buyer  $\tilde{n}$  induced trade on this platform. As a consequence, more than 2 mutations are needed to reach  $\omega_j^*$ .

In case (b), under asynchronous learning only one trader has the opportunity to revise its platform choice in any given period. Suppose one seller and buyer  $\tilde{n}$  switch to platform j by mutation. Suppose furthermore that in the subsequent

rounds only sellers and buyers at platform *i* receive the opportunity to revise their decision (this happens with strictly positive probability due to B1 and B2). If  $\frac{1}{|M|-1}d_{\tilde{n}}(\frac{\beta_j c}{1-f_j})(\beta_j-1) \geq D_{N\setminus\{\tilde{n}\}}\left(\frac{\beta_i c}{1-f_i}\right)(\beta_i-1)$  it follows that  $\frac{1}{|M_j|}d_{\tilde{n}}(\frac{\beta_j}{1-f_j})(\beta_j-1) \geq \frac{1}{|M|-|M_j|}D_{N\setminus\{\tilde{n}\}}\left(\frac{\beta_i}{1-f_i}\right)(\beta_i-1)$  for all  $M_j$  with  $1 \leq |M_j| \leq |M|-1$ . Hence, sellers prefer platform *j* whenever it is active and there are at least one and less than |M| sellers already there. Thus, there is a positive probability path with just two mutations from  $\omega_i^*$  to  $\omega_j^*$  where first all sellers move to platform *j* and subsequently all buyers switch to *j* as it is the only active platform. Hence, the condition displayed in the Lemma is sufficient for the stochastic stability of  $\omega_j^*$ . To see that it is also necessary, suppose it is not fulfilled. Then a seller at platform *i* prefers to stay there if all other sellers are at platform *j* together with any buyer  $\tilde{n}$ . Under asynchronous learning this implies that at least a third mutation is needed to reach  $\omega_j^*$ , which implies that this state cannot be stochastically stable by Theorem 2(a).

**A.3.** Proof of Proposition 10. Note first that Lemma 5 (a) implies that if  $\frac{\beta_i c}{1-f_i} < \frac{\beta_j c}{1-f_i}, \omega_j^*$  is stochastically stable iff

$$d\left(\frac{\beta_j c}{1-f_j}\right)(\beta_j - 1) \ge d\left(\frac{\beta_i c}{1-f_i}\right)(\beta_i - 1).$$
(\*)

Proof of i) Assume to the contrary for all platform characteristics in the support of  $\sigma_i^*$  and  $\sigma_j^*$ ,  $\beta = 1 + \delta$ . Denote by  $\overline{f_i}$  the highest fee of a platform in the support of  $\sigma_i^*$ , and by  $\overline{f_j}$  the highest fee of a platform in the support of  $\sigma_j^*$ . Without loss of generality assume that  $\overline{f_j} \ge \overline{f_i}$ . We can distinguish between three cases:

a)  $\overline{f}_j > \overline{f}_i$ : Condition (\*) shows that full coordination on platform  $\overline{s}_j = (1 + \delta, \overline{f}_j)$ is not stochastically stable vis a vis any platform characteristics in the support of  $\sigma_i^*$ . Hence, this strategy earns designer j zero profits, and since it is assumed to be in the support of j's equilibrium strategy, j's equilibrium profits would be zero. But jcould always guarantee himself a strictly positive profit by playing the same (possibly mixed) strategy as i. Hence, case (a) is inconsistent with Nash equilibrium.

b)  $\overline{f}_i = \overline{f}_j > f_{\min}$ . Condition (\*) shows that full coordination on platform  $\overline{s}_i = (1+\delta, \overline{f}_i)$  is not stochastically stable vis a vis any platform characteristics in the support of  $\sigma_j^*$  but platform  $\overline{s}_j = (1+\delta, \overline{f}_j)$ . Furthermore, Lemmata 2 and 3 imply that  $\mu^*(\omega_i^*) = \mu^*(\omega_j^*) = \frac{1}{2}$  if  $\overline{s}_i$  is chosen by *i* and  $\overline{s}_j$  is chosen by *j*. Therefore,

$$\pi_{D,i}(\overline{s}_i, \sigma_j^*) = \sigma_j^*(\overline{s}_j) \frac{1}{2} \overline{f}_i \frac{(1+\delta)c}{1-\overline{f}_i} |N| d\left(\frac{(1+\delta)c}{1-\overline{f}_i}\right)$$

But choosing the alternative platform design  $s'_i$  with  $f'_i = \overline{f}_j - \gamma$ , and  $\beta'_i = 1 + \delta$ implies that  $\frac{1+\delta}{1-f'_i}c < \frac{1+\delta}{1-\overline{f}_j}c$  and  $d\left(\frac{1+\delta}{1-\overline{f}_j}c\right)\delta < d\left(\frac{1+\delta}{1-f'_i}c\right)\delta$ . Hence, again by (\*)  $\mu^*(\omega_i^*) = 1$  if  $s'_i$  is chosen by i and  $\overline{s}_j$  is chosen by j, which yields

$$\pi_{D,i}(s'_i,\sigma^*_j) \ge \sigma^*_j(\overline{s}_j)(\overline{f}_i-\gamma)\frac{(1+\delta)c}{1-\overline{f}_i+\gamma}|N|d\left(\frac{(1+\delta)c}{1-\overline{f}_i+\gamma}\right).$$

If the grid of F is fine enough, i.e. if  $\gamma$  is small enough, this implies  $\pi_{D,i}(s'_i, \sigma^*_j) > \pi_{D,i}(\overline{s}_i, \sigma^*_j)$ , a contradiction with Nash equilibrium.

c)  $\overline{f}_i = \overline{f}_j = f_{\min}$  - both designers choose the platform characteristics  $\overline{s}_i = \overline{s}_j = (1 + \delta, f_{\min})$  for sure. Then Condition (\*) guarantees the existence of a  $\beta'_j > 1 + \delta$  and a  $f'_j > f_{\min}$  such that platform j is stochastically stable vis a vis  $\overline{s}_j$  if the grid F is sufficiently fine, i.e. if  $\gamma$  is sufficiently small. Furthermore, if  $\varepsilon_p$  is not much larger than 1, designer j's profits from full coordination on his platform with design  $s'_j$ , i.e.  $|N| f'_j \frac{\beta'_j c}{1 - f'_j} d\left(\frac{\beta'_j c}{1 - f'_j}\right)$ , is strictly larger than the respective profit from choosing  $\overline{s}_j$ , i.e.  $|N| \overline{f}_j \frac{(1+\delta)c}{1-\overline{f}_j} d\left(\frac{(1+\delta)c}{1-\overline{f}_j}\right)$ . It remains to show that no decrease in  $\mu^*(\omega_j^*)$  overcompensates this effect. To see this suppose that  $\beta'_j > \overline{\beta}_j = 1 + \delta$  in such a way that  $d(p'_j)(\beta'_j - 1) > |M| |N| d(p_i) \delta$  (feasible if  $\delta$  is sufficiently small). Then sellers prefer platform j with characteristics  $\overline{s}_i$ . For a learning dynamics with independent inertia and with |M| = |N| this establishes symmetry of the transition matrix P such that  $\mu^*(\omega_i^*) = \mu^*(\omega_j^*) = 1/2$ . Therefore, choosing in this case  $\beta'_j > 1 + \delta$  and  $f'_j > \overline{f}_j$  does not reduce  $\mu^*(\omega_j^*)$  while it strictly increases revenue in  $\omega_j^*$ . Hence,  $\overline{s}_j$  with  $\overline{\beta}_j = 1 + \delta$  and  $\overline{f}_i = f_{min}$  can not be a best response to  $\overline{s}_i = (1 + \delta, f_{min})$ .

Proof of ii) Follows immediately from i). ■

#### B. ROBUSTNESS OF THE RESULTS

The results in the paper have been derived under two crucial assumptions. First, designers are assumed to be rational while traders are not (asymmetric rationality). Even though this assumption seems to be justified in a wide range of applications, one might be interested in the robustness or our results with respect to the (bounded) rationality of designers. We will discuss the case of learning designers in section B.1. Second, we assumed that sellers have a constant-returns-to-scale technology. In section B.2 we will analyze an example with decreasing returns that illustrates the robustness of our findings.

**B.1.** Boundedly Rational Designers. To account for learning designers, we have to extend the state space by the feasible design configurations, and we have to redefine the (unperturbed learning process).

The state space is given by  $\Omega = \{1,2\}^n \times \{1,2\}^m \times S^2$ . A state  $\omega \in \Omega$  denotes the location of buyers and sellers and the design of both platforms. Traders learn as specified in Assumption A. The learning process of designers is defined as follows.

**Assumption C** A designer who gets the opportunity to revise, observes the revenues and designs of platforms in the last period. If the platforms have different designs, he chooses the design which has led to a higher revenue (*Imitation*). In case of identical revenues designers randomize their choice, with both designs chosen with strictly positive probability. Choice probabilities may depend on the outcomes (or designs) but not on the platform's name. If designs are identical and a designer obtains zero revenues he randomizes over all possible design alternatives in the next round (*Innovation*).<sup>19</sup>

We further assume B1 and B2 (on the enlarged state space and for three instead of two different types of players) as detailed in the paper.<sup>20</sup> To illustrate the robustness of our results with respect to boundedly rational designers, we prove the counterpart of Lemma 1 in the modified learning model.

**Lemma 11.** Assume A, B1, B2, and C. Then the absorbing sets are all sets of the form  $A_{i,\beta,f} = \{\omega \in \Omega | N_i(\omega) = N, M_i(\omega) = M, \beta_i = \beta, f_i = f\} \forall \beta > 1, f \in F$ , and i = 1, 2.

**Proof.** We proceed in two steps. First, we show that  $A_{i,\beta,f}$  is indeed absorbing for any  $i, f, \text{ and } \beta > 1$ . Second, we show that there is always a positive probability path from any  $\omega \in \Omega \setminus \bigcup_{i,f,\beta>1} A_{i,\beta,f}$  to an  $\omega' \in \bigcup_{i,f,\beta>1} A_{i,\beta,f}$ .

Fix any i, f and  $\beta > 1$ , and consider any  $\omega \in A_{i,\beta,f}$ . As  $\beta > 1$  buyers and sellers receive strictly positive payoffs at i, and no type of trader wants to switch to the inactive platform j. By Assumption C, designer i will not change the design of his platform while designer j will innovate (randomize over S) if  $s_i = s_j$  in the last period, and he will imitate (choose  $s_j = s_i$ ) otherwise. Hence, once the system reaches a state in  $A_{i,\beta,f}$  buyers, sellers, and designer i will not alter their choices and designer j will alternate between  $s_j = s_i$  and a random draw out of S - the system will not leave  $A_{i,\beta,f}$ .

Now fix a state  $\omega \in \Omega \setminus \bigcup_{i,f,\beta>1} A_{i,\beta,f}$  and denote the designs of platform i and jin this state by  $s_i = (\beta_i, f_i)$  and  $s_j = (\beta_j, f_j)$ , respectively. We distinguish four cases. If (i)  $\beta_i, \beta_j > 1$  or (ii)  $\beta_i > 1$  and  $\beta_j \leq 1$ , it follows from the proof of Lemma 1 that the unperturbed dynamics will reach the monomorphic state  $\omega_i^*$  (with  $\beta_i > 1$ ) with positive probability from any non-monomorphic state or any monomorphic state  $\omega_j^*$ with  $\beta_j \leq 1$ . By B1 and B2, there is positive probability that only designers with highest revenues receive revision opportunities (and do not change design) until a state  $\omega'$  with  $N_i(\omega') = N, M_i(\omega') = M$  is reached. If (iii)  $\beta_i = \beta_j = 1$ , the proof of Lemma 1 shows that a cross-state will be reached with positive probability. In such a state designers randomize, thus there is a positive probability path to a cross-state with  $\beta_i > 1$ , and a subsequent positive probability path to a state in  $A_{i,\beta,f}$  (due to randomizing traders). Finally consider (iv)  $\beta_i, \beta_j < 1$ . Traders randomize and a cross-state with  $\beta_i > 1$  can be reached. As traders continue to randomize, the dynamics reaches a state in  $A_{i,\beta_i,f_i}$  with positive probability.  $\blacksquare$ 

<sup>&</sup>lt;sup>19</sup>For simplicity, we assume that designers randomize over S with full support. Our results would carry over to more realistic treatments where designers use more sophisticated innovation methods.

<sup>&</sup>lt;sup>20</sup>This specification also allows for different learning speeds for traders and designers, respectively. Our model covers, for instance, the likely situation that buyers and sellers revise with a larger probability than designers.

ON THE EVOLUTION OF MARKET INSTITUTIONS: THE PLATFORM DESIGN PARADOX 29

Hence, in every absorbing set all traders are located at a platform i with  $\beta_i > 1$  while designers of the other platform randomize over  $S = B \times F$  (i.e. unsuccessfully innovate). As transition costs between any two absorbing sets are 1 (starting in a state in  $A_{i,\beta,f}$  just one experiment by designer i is required to end up in any other  $A_{i',\beta',f'}$ ) all absorbing states characterized in Lemma 11 are also stochastically stable. This establishes Theorem 9 for boundedly rational designers.

We conclude that boundedly rational platform designers exhibit qualitatively the same behavior as rational ones: Platform competition forces them to introduce only non-market clearing platforms.

**B.2.** Decreasing Returns to Scale. Consider the following example. Two identical sellers produce with costs given by  $c(q) = \frac{1}{2}q^2$ . For given prices  $(p_i)$  and fees  $(f_i)$  at a platform i, their profit is  $\pi_{m,i}(q, p_i) = (1 - f_i)p_iq - \frac{1}{2}q^2$  and maximization leads to the supply function  $s(p_i) = (1 - f_i)p_i$ . Two identical buyers, each with income of one unit, consume q units of the commodity traded at the platforms and x units of a second commodity which price is normalized to 1. The buyers' utility is given by  $\pi_n(q, x) = 2\sqrt{q} + x$  and utility maximization for a given price  $p_i$  at the respective platform yields the buyer's demand function  $d(p_i) = 1/p_i^2$ . Equating demand and supply gives the market clearing price at platform i in state  $\omega$   $p_i^*(\omega) = r_i(\omega)^{1/3}(1 - f_i)^{-1/3}$  (with  $r_i(\omega) = \frac{|N_i(\omega)|}{|M_i(\omega)|}$ ). Traders' and designers' profits depend on state and design and are calculated the same way as before. For our purposes it suffices to note that sellers are not rationed whenever  $\beta_i \leq 1$  and their corresponding profit  $\pi_{m,i}(M_i(\omega), N_i(\omega), s_i) = \frac{1}{2}(1 - f_i)^{4/3}\beta_i^2 r_i^{2/3}$  is increasing in  $\beta_i$  and decreasing in  $f_i$ . Analogously, sellers are rationed for  $\beta_i > 1$  and profits amount to  $\pi_{m,i}(M_i(\omega), N_i(\omega), s_i) = (1 - f_i)^{4/3} \frac{1}{\beta_i} r_i^{2/3} (1 - \frac{1}{2\beta_i^3})$  which is also monotonically decreasing in  $f_i$  but reaches a (global) maximum at  $\beta_i = 2^{1/3} \cdot 2^{1}$ 

Full coordination on any platform is a singleton-absorbing set (both types of traders get strictly positive profits on any active platform). Analogously to Lemma 1, it is easy to see that these states are the only absorbing sets. Moreover, it can be shown that these monomorphic states are also stochastically stable for a wide range of design configurations. The designers' profits, though, depend not only on the support of the limit invariant distribution  $\mu^*$ , but also on its weights for the different (monomorphic) states. Hence, this setting can not be analyzed with the tools discussed before. To obtain  $\mu^*$  in this example, we can use Lemma 3.1 from Freidlin and Wentzell ([20])<sup>22</sup>. They derive the limit invariant distribution  $\mu^*$  from an analysis of graphs in the state space  $\Omega$  that are induced by the stationary Markov-process P as follows. Fix an  $\omega \in \Omega$ . An  $\omega$ - tree T is a spanning tree in  $\Omega$  such that for every vertex  $\omega' \neq \omega$  there exists a unique directed path from  $\omega'$  to  $\omega$ . Let  $\mathcal{T}_{\omega}$  be the set of all  $\omega$ - trees and define  $q_{\omega} \equiv \sum_{T \in \mathcal{T}_{\omega}} \prod_{(\omega', \omega'') \in T} P(\omega', \omega'')$  (i.e.  $q_{\omega}$  is the product of all transition probabilities on a given  $\omega$ -tree summed over all  $\omega$ -trees).

<sup>&</sup>lt;sup>21</sup>We assume for simplicity from now on that  $\delta$  is such that  $2^{1/3} \in B$ .

<sup>&</sup>lt;sup>22</sup>For a detailed explanation see Kandori et al. [26] or Young [43].

Then Lemma 3.1 of Freidlin and Wentzell [20] states that  $\mu^*(\omega) = \frac{q_\omega}{\sum_{\omega \in \Omega} q_\omega}$ . In the limit of small mutation probability ( $\varepsilon \to 0$ )  $\mu^*(\omega)$  is determined by those  $\omega$ -trees that imply the smallest possible number of mutations necessary to form a spanning tree in  $\Omega$ . Hence, we will restrict ourselves to those minimal-cost trees.

For expositional ease we further specify the learning model and assume independent inertia within types. That is, in every round any seller  $m \in M$  is allowed to revise his location decision with probability  $\rho_S \in ]0,1[$  while every buyer is allowed to revise with probability  $\rho_B \in ]0,1[$ .

Then, a minimal-cost tree in  $\mathcal{T}_{\omega_i^*}$  has edges as depicted in the following table.

$\omega'$	$P_{\omega',\omega_i^*}$
(0,0)	$\varepsilon^2 P_{(1,1),\omega_i^*}$
(0,1)	$\varepsilon(1-\rho_B)P_{(1,1),\omega_i^*}$
(0,2)	$arepsilon ho_S$
(1,0)	$\varepsilon(1-\rho_S)P_{(1,1),\omega_i^*}$
(1,1)	$P_{(1,1),\omega_i^*}$
(1,2)	$ ho_S$
(2,0)	$\varepsilon  ho_B$
(2,1)	$ ho_B$

The respective table for  $\omega_j^*$  can be derived by a permutation of indices B and S. Hence,  $\frac{\mu^*(\omega_j^*)}{\mu^*(\omega_i^*)} = \rho_B^2 \rho_S^2 (1 - \rho_B) (1 - \rho_S) \left( P_{(1,1),\omega_j^*} \right)^4 / \rho_B^2 \rho_S^2 (1 - \rho_B) (1 - \rho_S) \left( P_{(1,1),\omega_i^*} \right)^4$ . Note that this expression is continuous in  $\rho_S$  and  $\rho_B$ . This leads to the following useful result (as sellers and buyers are identical profits only depend on the *number* of sellers and buyers at a platform).

**Lemma 12.** Suppose  $\pi_{m,i}(1,1,s_i) > \pi_{m,j}(1,1,s_j)$ . Then for every  $\varepsilon > 0$  there is a  $\overline{\rho_S} < 1$  such that  $\mu^*(\omega_i^*) < \varepsilon$  for all  $\rho_S > \overline{\rho_S}$ .

**Proof.** First recall that the monomorphic states are the only absorbing sets. Moreover, it is easy to see that  $\omega_i^*$  is the only stochastically stable state if and only if  $\pi_{m,i}(1,1,s_i) > \pi_{m,j}(1,1,s_j)$  and  $\pi_{n,i}(1,1,s_i) > \pi_{n,j}(1,1,s_j)$  (as it then needs more than 2 mistakes to get from  $\omega_i^*$  to  $\omega_j^*$ ). If  $\pi_{m,i}(1,1,s_i) > \pi_{m,j}(1,1,s_j)$  and  $\pi_{n,i}(1,1,s_i) = \pi_{n,j}(1,1,s_j)$ ,  $P_{(1,1),\omega_i^*} = \rho_S(1-\rho_B)\rho_B + \frac{1}{2}\rho_B(1-\rho_S)\rho_S + \frac{1}{2}\rho_B\rho_S$  while  $P_{(1,1),\omega_j^*} = \frac{1}{2}\rho_B(1-\rho_S)\rho_S$ . Hence,  $P_{(1,1),\omega_j^*}/P_{(1,1),\omega_i^*}$  approaches zero if  $\rho_S \to 1$  such that  $\lim_{\rho_S \to 1} \mu^*(\omega_j^*) = 0$ .

If  $\pi_{m,i}(1,1,s_i) > \pi_{m,j}(1,1,s_j)$  and  $\pi_{n,i}(1,1,s_i) < \pi_{n,j}(1,1,s_j)$ ,  $P_{(1,1),\omega_i^*} = \rho_S(1-\rho_B)\rho_B$  while  $P_{(1,1),\omega_j^*} = \rho_B(1-\rho_S)\rho_S$ . Hence,  $P_{(1,1),\omega_j^*}/P_{(1,1),\omega_i^*}$  again approaches zero if  $\rho_S \to 1$  such that  $\lim_{\rho_S \to 1} \mu^*(\omega_j^*) = 0$ .

Intuitively, if sellers learn much faster then buyers, only the platform that offers higher revenues to sellers will survive with a positive probability if both platforms are active. This induces the following strict Nash-Equilibrium.

**Proposition 13.** There exists a  $\overline{\rho_S} < 1$  such that, for all  $\rho_S > \overline{\rho_S}$ , the platform profile  $(s_i^*, s_j^*)$  with  $f_i^* = f_j^* = f_{\min}$  and  $\beta_i^* = \beta_i^* = 2^{1/3}$  is a strict Nash equilibrium.

**Proof.** As seller's profits decrease in  $f_i$  and reach their global maximum in  $\beta = 2^{1/3}$  it is clear that  $\pi_{m,i}(1,1,s_i) < \pi_{m,i}(1,1,s_i^*)$  for any  $s_i \neq s_i^*$ . But then Lemma 12 indicates that for any  $\varepsilon > 0$  there is a  $\overline{\rho_S}$  such that  $\mu^*(\omega_i^*, s_i', s_j^*) < \varepsilon$  for all  $\rho_S > \overline{\rho_S}$ . Hence, if  $\varepsilon$  is chosen small enough, we obtain  $\pi_{D,i}(s_i', s_j^*) < \pi_{D,i}(s_i^*, s_j^*)$  for all  $\rho_S > \overline{\rho_S}$ ; thus  $(s_i^*, s_j^*)$  is a strict Nash equilibrium.

It can be shown that, for  $\rho_s$  large enough,  $\beta_i \leq 1$  is not chosen by any designer in any pure strategy equilibrium. Moreover, in any mixed strategy equilibrium  $(\sigma_1^*, \sigma_2^*)$ there is a least one designer *i* where  $s_i \in C(\sigma_i^*)$  implies that  $\beta_i > 1.^{23}$ 

The previous proposition shows that also for decreasing returns to scale competition between market designers might lead to the design of non-market clearing institutions. Note that the condition on  $\rho_S$  is a sufficient, but not a necessary one. For a smaller  $\rho_S$  we cannot characterize the limit invariant distribution, and hence do not know the equilibrium behavior of the designers. So our main result might still hold even for  $\rho_S < \overline{\rho_S}$ , but this depends on the details of the demand and supply conditions.

### C. A MARKOV CHAINS DICTIONARY

- A finite Markov Chain with stationary transition probabilities is a pair  $(\Omega, P)$ of a finite state space  $\Omega$  and a transition probability matrix  $P : \Omega \times \Omega \to [0, 1]$ where  $P(\omega, \omega')$  is the probability that the state will be in  $\omega' \in \Omega$  in period t+1given that it was in  $\omega \in \Omega$  in period t; thus  $\sum_{\omega' \in \Omega} P(\omega, \omega') = 1$  for each  $\omega \in \Omega$ . Since in this paper we restrict ourselves to finite Markov chains with stationary transition probabilities, we will drop both adjectives and refer to them simply as Markov chains. A Path of such a Markov chain is a mapping from a countable (time) set (in our case  $Z^+$ ) to the state space that depicts the evolution of the process  $\omega(\cdot) : Z^+ \to \Omega$ .
- An Absorbing Set (also Recurrent Communication Class or Limit Set)  $A \subseteq \Omega$  is a minimal subset of  $\Omega$  that once entered is never abandoned (i.e.  $P(\omega, \omega') = 0$  $\forall \omega \in A$  and  $\omega' \notin A$ ). States that do not belong to any absorbing set are called *transient*. Note that any finite Markov chain has at least one absorbing set.
- An Absorbing State (or a Singleton Recurrent Communication Class) is a state  $a \in \Omega$  with P(a, a) = 1 (and as a consequence  $P(a, \omega) = 0 \ \forall \omega \in \Omega \setminus \{a\}$ ).
- Every Markov chain induces (a set of) Invariant Distributions  $\mu : \Omega \to [0, 1]$ with  $\sum_{\omega \in \Omega} \mu(\omega) = 1$  (i.e.  $\mu \in \Delta(\Omega)$ ) and  $\mu \cdot P = \mu$  (i.e. the probability distribution  $\mu$  over  $\Omega$  is invariant with respect to P). Every absorbing set  $A \subseteq \Omega$  corresponds to exactly one invariant distribution with support A (i.e.

 $<sup>^{23}</sup>$ A proof of these claims and more detailed exposition of the material discussed in the Appendix is available on request.

 $\mu(\omega) > 0$  if and only if  $\omega \in A$ ). The set of all invariant distributions of a Markov Process is the convex hull of the invariant distributions of all its absorbing sets.

- By the *Ergodic Theorem* an invariant distribution which is induced by an absorbing set  $A \subseteq \Omega$  describes the time-average behavior of the system once it reached A. I.e.  $\mu(\omega)$  equals the (long-run) fraction of time that the system is in state  $\omega \in A$ , along almost any path of the chain that reaches a state in A.
- A Markov chain is *ergodic* if it has a unique absorbing set. Note that the invariant distribution of an ergodic Markov process is unique and thus depicts the time-average behavior independent of the initial conditions (the initial probability distribution over states). If the chain is not ergodic, then several invariant distributions exist, describing the long-run behavior along different sample paths, i.e. the prediction depends on the initial conditions.
- A Markov chain is *irreducible* if it is ergodic and the unique absorbing set coincides with the state space  $\Omega$ .
- The matrix P describes the one-step transition probabilities. Let  $P^k = P \cdot \stackrel{k}{\ldots} P$ describe the k-step transition probabilities. The period of a state  $\omega$  is the greatest common divisor of all integers k such that  $P^k(\omega) > 0$ . All states in the same absorbing set have the same period. An absorbing set is *aperiodic* if that period is one. An example of a non-aperiodic absorbing set would be one containing a deterministic, non-trivial cycle, i.e.  $\{\omega_1, \omega_2, ..., \omega_n\}$  with  $n \ge 2$ such that  $P(\omega_i, \omega_{i+1}) = P(\omega_n, \omega_1) = 1$  for all  $i \in \{1, ..., n-1\}$ . Note that a sufficient condition for the aperiodicity of an absorbing set A is that there is an  $\omega \in A$  such that  $P(\omega, \omega) > 0$ , i.e. that the Markov process exhibits some inertia. Any absorbing state is obviously aperiodic.
- By the Fundamental Theorem of Markov Chains the invariant distribution  $\mu$ induced by an aperiodic absorbing set  $A \subseteq \Omega$  describes the long-run probability that the system will be in state  $\omega$ , given that the path reaches A, i.e.  $\mu(\omega) = \lim_{T \to \infty} \nu P^T$  for every  $\omega \in A$  and all probability distributions over states  $\nu \in \Delta(\Omega)$  whose support is contained in A.
- A Perturbed Markov Chain  $(\Omega, P^{\varepsilon})$  is a Markov chain  $(\Omega, P)$  and a parameter  $\varepsilon \geq 0$  such that (i)  $(\Omega, P^{\varepsilon})$  is aperiodic and irreducible for every  $\varepsilon > 0$ , (ii) all transition probabilities  $P^{\varepsilon}(\omega, \omega')$  are continuous in  $\varepsilon$  with  $P^{0} = P$ , and (iii) whenever  $P^{\varepsilon}(\omega, \omega') > 0$  for some  $\varepsilon > 0$ , this implies that there is an  $r \geq 0$  such that  $\infty > \lim_{\varepsilon \to 0} \varepsilon^{-r} P^{\varepsilon}(\omega, \omega') > 0$ . Such perturbations are also referred to as regular, and r is referred to as the cost of transition from  $\omega$  to  $\omega'$ . Typically, the dynamics is formulated in such a way that r can be interpreted as the number of experiments or mistakes that agents must make for the transition to happen with positive probability.

- Any perturbed Markov chain is irreducible, hence its invariant distribution μ<sup>ε</sup> is unique. Moreover, the *Limit Invariant Distribution* μ<sup>\*</sup> ≡ lim<sub>ε→0</sub> μ<sup>ε</sup> of a Markov chain (Ω, P) exists and is an invariant distribution of (Ω, P) (see e.g. Theorem 4(i) in [43]). This implies that only absorbing sets can be in the support of μ<sup>\*</sup>. Intuitively, the limit invariant distribution singles out a stable prediction of the unperturbed dynamics, in the sense that, for any ε > 0 small enough, the process approximates the one described by μ<sup>\*</sup> in the long run.
- A state in the support of  $\mu^*$  (i.e.  $\omega \in \{\omega \in \Omega | \mu^*(\omega) > 0\}$ ) is called a *Stochastically Stable State*. Two states in the same absorbing set are either both stochastically stable or not.
- For just two absorbing sets of the unperturbed learning process the set of stochastically stable states can be characterized as follows.<sup>24</sup> Suppose X and Y are the only absorbing sets of a Markov process  $(\Omega, P)$ . Denote by  $\gamma(X, Y)$  the minimal number of experiments necessary for a direct transition from X to Y, i.e. a positive probability path starting in an element of X and leading to an element in Y, which does not go through any other absorbing set. Then the states in X are stochastically stable if and only if  $\gamma(X, Y) \ge \gamma(Y, X)$ .

<sup>&</sup>lt;sup>24</sup>This is a special case of the results developed by Kandori et al. [26] and Young [43]. Detailed overviews can be found e.g. in Ellison [13], Fudenberg and Levine [19] or Samuelson [39].

# **CESifo Working Paper Series**

(for full list see www.cesifo-group.de)

- 1948 C. Mirjam van Praag and Bernard M. S. van Praag, The Benefits of Being Economics Professor A (and not Z), March 2007
- 1949 Astrid Hopfensitz and Frans van Winden, Dynamic Choice, Independence and Emotions, March 2007
- 1950 Guglielmo Maria Caporale and Luis A. Gil-Alana, A Multivariate Long-Memory Model with Structural Breaks, March 2007
- 1951 Mattias Ganslandt and Keith E. Maskus, Wholesale Price Discrimination and Parallel Imports, March 2007
- 1952 Michela Redoano, Fiscal Interactions Among European Countries. Does the EU Matter?, March 2007
- 1953 Stefan C. Wolter, Rémy Hübschi and Matthias Müller, Push or Pull? An Empirical Analysis of the Demand for Individual Project Grants from the Swiss National Science Foundation, March 2007
- 1954 Scott Alan Carson, African-American and White Inequality in the American South: Evidence from the 19<sup>th</sup> Century Missouri State Prison, March 2007
- 1955 Peter Egger, Marko Koethenbuerger and Michael Smart, Do Fiscal Transfers Alleviate Business Tax Competition? Evidence from Germany, March 2007
- 1956 Panu Poutvaara and Lars-H. R. Siemers, Smoking and Social Interaction, March 2007
- 1957 Stephan Danninger and Fred Joutz, What Explains Germany's Rebounding Export Market Share?, March 2007
- 1958 Stefan Krasa and Mattias Polborn, Majority-efficiency and Competition-efficiency in a Binary Policy Model, March 2007
- 1959 Thiess Buettner and Georg Wamser, Intercompany Loans and Profit Shifting Evidence from Company-Level Data, March 2007
- 1960 Per Pettersson-Lidbom and Mikael Priks, Behavior under Social Pressure: Empty Italian Stadiums and Referee Bias, April 2007
- 1961 Balázs Égert and Carol S. Leonard, Dutch Disease Scare in Kazakhstan: Is it real?, April 2007
- 1962 Paul De Grauwe and Pablo Rovira Kaltwasser, Modeling Optimism and Pessimism in the Foreign Exchange Market, April 2007

- 1963 Volker Grossmann and Thomas M. Steger, Anti-Competitive Conduct, In-House R&D, and Growth, April 2007
- 1964 Steven Brakman and Charles van Marrewijk, It's a Big World After All, April 2007
- 1965 Mauro Ghinamo, Paolo M. Panteghini and Federico Revelli, FDI Determination and Corporate Tax Competition in a Volatile World, April 2007
- 1966 Inés Macho-Stadler and David Pérez-Castrillo, Optimal Monitoring to Implement Clean Technologies when Pollution is Random, April 2007
- 1967 Thomas Eichner and Ruediger Pethig, Efficient CO<sub>2</sub> Emissions Control with National Emissions Taxes and International Emissions Trading, April 2007
- 1968 Michela Redoano, Does Centralization Affect the Number and Size of Lobbies?, April 2007
- 1969 Christian Gollier, Intergenerational Risk-Sharing and Risk-Taking of a Pension Fund, April 2007
- 1970 Swapan K. Bhattacharya and Biswa N. Bhattacharyay, Gains and Losses of India-China Trade Cooperation – a Gravity Model Impact Analysis, April 2007
- 1971 Gerhard Illing, Financial Stability and Monetary Policy A Framework, April 2007
- 1972 Rainald Borck and Matthias Wrede, Commuting Subsidies with two Transport Modes, April 2007
- 1973 Frederick van der Ploeg, Prudent Budgetary Policy: Political Economy of Precautionary Taxation, April 2007
- 1974 Ben J. Heijdra and Ward E. Romp, Retirement, Pensions, and Ageing, April 2007
- 1975 Scott Alan Carson, Health during Industrialization: Evidence from the 19<sup>th</sup> Century Pennsylvania State Prison System, April 2007
- 1976 Andreas Haufler and Ian Wooton, Competition for Firms in an Oligopolistic Industry: Do Firms or Countries Have to Pay?, April 2007
- 1977 Eckhard Janeba, Exports, Unemployment and the Welfare State, April 2007
- 1978 Gernot Doppelhofer and Melvyn Weeks, Jointness of Growth Determinants, April 2007
- 1979 Edith Sand and Assaf Razin, The Role of Immigration in Sustaining the Social Security System: A Political Economy Approach, April 2007
- 1980 Marco Pagano and Giovanni Immordino, Optimal Regulation of Auditing, May 2007
- 1981 Ludger Woessmann, Fundamental Determinants of School Efficiency and Equity: German States as a Microcosm for OECD Countries, May 2007

- 1982 Bas Jacobs, Real Options and Human Capital Investment, May 2007
- 1983 Steinar Holden and Fredrik Wulfsberg, Are Real Wages Rigid Downwards?, May 2007
- 1984 Cheng Hsiao, M. Hashem Pesaran and Andreas Pick, Diagnostic Tests of Cross Section Independence for Nonlinear Panel Data Models, May 2007
- 1985 Luis Otávio Façanha and Marcelo Resende, Hierarchical Structure in Brazilian Industrial Firms: An Econometric Study, May 2007
- 1986 Ondřej Schneider, The EU Budget Dispute A Blessing in Disguise?, May2007
- 1987 Sascha O. Becker and Ludger Woessmann, Was Weber Wrong? A Human Capital Theory of Protestant Economic History, May 2007
- 1988 Erkki Koskela and Rune Stenbacka, Equilibrium Unemployment with Outsourcing and Wage Solidarity under Labour Market Imperfections, May 2007
- 1989 Guglielmo Maria Caporale, Juncal Cunado and Luis A. Gil-Alana, Deterministic versus Stochastic Seasonal Fractional Integration and Structural Breaks, May 2007
- 1990 Cláudia Costa Storti and Paul De Grauwe, Globalization and the Price Decline of Illicit Drugs, May 2007
- 1991 Thomas Eichner and Ruediger Pethig, Pricing the Ecosystem and Taxing Ecosystem Services: A General Equilibrium Approach, May 2007
- 1992 Wladimir Raymond, Pierre Mohnen, Franz Palm and Sybrand Schim van der Loeff, The Behavior of the Maximum Likelihood Estimator of Dynamic Panel Data Sample Selection Models, May 2007
- 1993 Fahad Khalil, Jacques Lawarrée and Sungho Yun, Bribery vs. Extortion: Allowing the Lesser of two Evils, May 2007
- 1994 Thorvaldur Gylfason, The International Economics of Natural Resources and Growth, May 2007
- 1995 Catherine Roux and Thomas von Ungern-Sternberg, Leniency Programs in a Multimarket Setting: Amnesty Plus and Penalty Plus, May 2007
- 1996 J. Atsu Amegashie, Bazoumana Ouattara and Eric Strobl, Moral Hazard and the Composition of Transfers: Theory with an Application to Foreign Aid, May 2007
- 1997 Wolfgang Buchholz and Wolfgang Peters, Equal Sacrifice and Fair Burden Sharing in a Public Goods Economy, May 2007
- 1998 Robert S. Chirinko and Debdulal Mallick, The Fisher/Cobb-Douglas Paradox, Factor Shares, and Cointegration, May 2007
- 1999 Petra M. Geraats, Political Pressures and Monetary Mystique, May 2007

- 2000 Hartmut Egger and Udo Kreickemeier, Firm Heterogeneity and the Labour Market Effects of Trade Liberalisation, May 2007
- 2001 Andreas Freytag and Friedrich Schneider, Monetary Commitment, Institutional Constraints and Inflation: Empirical Evidence for OECD Countries since the 1970s, May 2007
- 2002 Niclas Berggren, Henrik Jordahl and Panu Poutvaara, The Looks of a Winner: Beauty, Gender, and Electoral Success, May 2007
- 2003 Tomer Blumkin, Yoram Margalioth and Efraim Sadka, Incorporating Affirmative Action into the Welfare State, May 2007
- 2004 Harrie A. A. Verbon, Migrating Football Players, Transfer Fees and Migration Controls, May 2007
- 2005 Helmuth Cremer, Jean-Marie Lozachmeur and Pierre Pestieau, Income Taxation of Couples and the Tax Unit Choice, May 2007
- 2006 Michele Moretto and Paolo M. Panteghini, Preemption, Start-Up Decisions and the Firms' Capital Structure, May 2007
- 2007 Andreas Schäfer and Thomas M. Steger, Macroeconomic Consequences of Distributional Conflicts, May 2007
- 2008 Mikael Priks, Judiciaries in Corrupt Societies, June 2007
- 2009 Steinar Holden and Fredrik Wulfsberg, Downward Nominal Wage Rigidity in the OECD, June 2007
- 2010 Emmanuel Dhyne, Catherine Fuss, Hashem Pesaran and Patrick Sevestre, Lumpy Price Adjustments: A Microeconometric Analysis, June 2007
- 2011 Paul Belleflamme and Eric Toulemonde, Negative Intra-Group Externalities in Two-Sided Markets, June 2007
- 2012 Carlos Alós-Ferrer, Georg Kirchsteiger and Markus Walzl, On the Evolution of Market Institutions: The Platform Design Paradox, June 2007